# SECURITIES AND EXCHANGE COMMISSION

# FORM 8-K/A

Current report filing [amend]

Filing Date: **2005-09-09** | Period of Report: **2005-09-01** SEC Accession No. 0001050502-05-000458

(HTML Version on secdatabase.com)

# **FILER**

# **GLOBAL AIRCRAFT SOLUTIONS, INC.**

CIK:854171| IRS No.: 841108499 | State of Incorp.:AZ | Fiscal Year End: 1231

Type: 8-K/A | Act: 34 | File No.: 000-28575 | Film No.: 051077463 SIC: 4581 Airports, flying fields & airport terminal services

Mailing Address 6900 S. PARK AVE. P.O. BOX 23009 TUCSON AZ 85706 Business Address 6900 S. PARK AVE. P.O. BOX 23009 TUCSON AZ 85706 5202943481

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549 FORM 8-K/A

(Amendment No. 1)

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of Earliest Event Reported): September 1, 2005

Global Aircraft Solutions, Inc.

\_\_\_\_\_

(Exact Name of Registrant as Specified in Charter)

Nevada

000 - 28575

84-1108499

----

\_\_\_\_\_

-----

(State of Incorporation)

(Commission File

(IRS Employer Identification No.)

Number)

P.O. Box 23009 Tucson, AZ 85734

(Address of principal executive offices)

(520) 294-3481

\_\_\_\_\_

(Registrant's telephone number, including area code)

Renegade Venture (NEV.) Corporation

(Former Name or Former Address if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

- |\_| Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- |\_| Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- | | Pre-commencement communications pursuant to Rule 14d-2(b) under the

Exchange Act (17 CFR 240.14d-2(b)).

|\_| Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-14(c).

ITEM 1.01. Entry into a Material Definitive Agreement.

On August 26, 2005, Global Aircraft Solutions, Inc. ("Global") together with BCI Aircraft Leasing ("BCI") formed a joint venture Delaware limited liability company called Jetglobal, LLC. This is a special purpose LLC formed to acquire and remarket commercial jet aircraft. BCI will be primarily responsible for the marketing aspects of Jetglobal while Global will be responsible for the technical, repair and maintenance aspects associated with remarketing purchased aircraft. Global invested an initial amount of \$1,125,000 for a 30% membership interest and BCI invested an initial amount of \$2,625,000 for a 70% membership interest in Jetglobal.

On September 2, 2005, Jetglobal, the limited liability company in which Global holds a 30% membership interest, entered into an agreement to acquire a fleet of 26 Boeing 737-200 aircraft from Jetran International. Of the 26 aircraft, 14 are parked and available for immediate re-sale or lease and 12 are still in service and currently under lease with Delta Airlines ("Delta"). The leases on the 12 aircraft currently in service with Delta Airlines are scheduled to expire between September 2006 and October 2007. All lease revenue generated by the 12 aircraft in service with Delta Airlines will devolve to Jetglobal until such time as the leases are either concluded or extended with Delta.

#### ITEM 9.01. Financial Statements and Exhibits

# (c) Exhibits

Exhibit No.	Document						
99.1	Operating Agreement of Jetglobal, LLC						
99.2	Press Release of Global Aircraft Solutions, Inc. issued September 1, 2005.						
99.3	Aircraft Sale & Purchase Agreement between Jetglobal, LLC and Jetran, LLC						

#### SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: September 9, 2005

Global Aircraft Solutions, Inc.

(Registrant)

By: /s/ John Sawyer

-----

Name: John Sawyer Title: President

# LIMITED LIABILITY COMPANY AGREEMENT OF JETGLOBAL, LLC

THIS AGREEMENT is made this 26th day of August, 2005, among Jetglobal, LLC, a Delaware limited liability company, Global Aircraft Solutions, Inc., a Nevada corporation, and BCI Aircraft Leasing, Inc., an Illinois corporation.

#### 1. RECITALS.

- 1.1 Formation. The Company is a limited liability company formed under the Act. A Certificate of Formation for the Company was filed with the Secretary of State of Delaware on August 25, 2005.
- 1.2 Intent. The Company and the Members wish to enter into this Limited Liability Company Agreement to provide for the manner in which the Company will acquire, own and sell its assets and carry out its business, to establish the means by which the Members will govern the Company and to set forth certain other agreements among the Members with respect to their rights and obligations relating to the Company and their ownership of the Company.
- 1.3 Purpose. The Company has been formed for the purpose of acquiring, directly or through acquisition of owner participant's interests in trusts, twenty six (26) Boeing 737-232 aircraft pursuant to a Purchase Agreement with Jetran.

Accordingly, the parties hereto agree as follows:

# 2. DEFINITIONS.

As used in this Agreement the following terms will have the following meanings.

- 2.1 "Act" will mean the Delaware Limited Liability Company Act, Del. Code Title 6, Chapter 18, in effect on the date of this Agreement and as amended from time to time.
- 2.2 "Adjusted Capital Account Deficit" will mean with respect to any Member, the deficit balance, if any, in such Member's Capital Account as of the end of the relevant tax year, after giving effect to the following adjustments:
  - (a) Credit to such Capital Account any amounts which such Member is obligated to restore or is deemed to be obligated to restore to the Company pursuant to the penultimate sentences of Regulations Sections 1.704-2(g)(1) and 1.704-2(g)(5); and

- (b) Debit to such Capital Account the items described in Sections 1.704-1 (b) (2) (ii) (d) (4), 1.704-1 (b) (2) (ii) (d) (5), and 1.7.04-1 (b) (2) (ii) (d) (6) of the Regulations.
- 2.3 "Aircraft" will mean, individually and collectively, each of the Boeing 737-232 aircraft to be acquired pursuant to the Purchase Agreement, and will include the airframes and Engines acquired pursuant to the Purchase Agreement.
- 2.4 "BCI" will mean BCI Aircraft Leasing, Inc., the address of which is One IBM Plaza, 330 North Wabash Avenue, Suite 2802, Chicago, Illinois 60611, and the facsimile number of which is (312) 329-1250.
- 2.5 "Capital Account" will mean, with respect to any Member, the Capital Account maintained for such Member in accordance with the provisions of paragraph 9.1.
- 2.6 "Capital Contribution" or "Capital Contributions" will mean the amount of cash and the net fair market value of any property contributed to the capital of the Company by the Members net of any debt related thereto assumed by the Company pursuant to this Agreement. The term "Capital Contributions" with respect to a Member will include (i) the contributions of such Member made pursuant to paragraph 8 and (ii) such Member's payments made to third party creditors of the Company after the Agreement Date with respect to Company obligations unless and until reimbursed by the Company.
- 2.7 "Code" will mean the Internal Revenue Code of 1986, as amended and in effect from time to time (or any corresponding provision of succeeding law).
- 2.8 "Company" will mean the limited liability company identified in the preamble paragraph of this Agreement.
- 2.9 "Company Accounting Year" will mean and refer to the accounting year of the Company ending December 31 of each calendar year or such shorter fiscal period during such year for which a relevant determination is being made under this Agreement.
- 2.10 "Company Minimum Gain" has the meaning ascribed to the term "Partnership Minimum Gain" in Regulations Section 1.704-2(d)(1)
- 2.11 "Disposition" is defined in the definition of "Gain or Loss on Disposition."
- 2.12 "Distribution" will mean any payment of cash or property by the Company to a Member other than payments with respect to loans.
- 2.13 "Engine" will mean each of the aircraft engines acquired as part of the Aircraft or as spare engines under the Purchase Agreement.
  - 2.14 "Gain" or "Loss" on "Disposition" will mean the gain or loss (as the

case may be) of the Company for federal income tax purposes, arising from a sale, exchange or other taxable disposition (including casualty or condemnation) of all or a portion of the Property or any other asset of the Company. Gain or loss resulting from any disposition of all or a portion of contributed property will be computed by reference to the agreed upon Gross Asset Value of the property disposed of (as adjusted for book purposes from time to time), notwithstanding that the adjusted tax basis of such property may differ from its agreed upon net fair market value.

- 2.15 "Global" will mean Global Aircraft Solutions, Inc., the address of which is 6901 South Park Avenue, Tucson, Arizona 85706, and the facsimile number of which is (520) 741-1430.
- 2.16 "Gross Asset Value" will mean, with respect to any asset, the adjusted basis of the asset for federal income tax purposes, adjusted as provided in paragraph 9.10.
- 2.17 "Income Tax Regulations" or "Regulations" will mean the final or temporary regulations promulgated from time to time under the Code or, if no final or temporary regulations with respect to a tax issue then are in effect, proposed regulations then in effect if approved by the Managers.
- 2.18 "Invested Capital" will mean with respect to each Member the Capital Contributions made from time to time to the Company by such Member, reduced by any distributions made to such Member pursuant to Section 10.1(b).
- 2.19 "Jetran" will mean Jetran International, Ltd., the address of which is 12400 Hwy 281 N. Suite 150, San Antonio, TX 78216, and the facsimile no. of which is (210) 495 7799.
- 2.20 "Manager" means the Manager or Managers elected or appointed in accordance with paragraph 5.1 or 5.5.
- 2.21 "Member" will mean BCI, Global and any other person or entity admitted as a member of the Company in accordance with this Agreement.
  - 2.22 "Member Loan" has the meaning set forth in Section 8.6.
- 2.23 "Member Non-recourse Debt" has the meaning for the term "Partner Non-recourse Debt" set forth in Regulations Section 1.704-2(b)(4).
- 2.24 "Member Non-recourse Debt Minimum Gain" means an amount, with respect to each Member Non-recourse Debt, equal to the Company Minimum Gain that would result if such Member Non-recourse Debt were treated as a Non-recourse Liability, determined in accordance with Regulations Section 1.704-2(i)(2).

2.25 "Member Non-Recourse Deductions" has the meaning for the term "Partner Non-Recourse Deductions" set forth in Regulations Section 1.704-2(i). The amount of Member Non-Recourse Deductions with respect to a Member Non-recourse Debt for a Company Accounting Year equals the excess, if any, (i) of the net increase, if any, in the amount of the Company Minimum Gain attributable to such Member Non-recourse Debt during such Company Accounting Year, over (ii) the aggregate amount of any Distributions during such year to the Member that bears the economic risk of loss for such Member Non-recourse Debt to the extent such Distributions are from proceeds of such Member Non-recourse Debt and are allocable to an increase in Member Non-recourse Debt Minimum Gain attributable to such Member Non-recourse Debt, determined according to the provisions of Regulations Section 1.704-2(i).

- 2.26 "Membership" or "Membership Interest" will mean the interest of each Member in the Company.
- 2.27 "Net Available Cash," with respect to any Company accounting period, will mean (i) the sum of all cash receipts of the Company during such year from all sources (including Capital Contributions, cash on hand at the beginning of the Company's accounting period to the extent not held in reserves, Distributions from any subsidiary and any funds released during such Company accounting period from cash reserves previously established), minus (ii) the sum of (a) Capital Receipts, (b) Net Loan Proceeds, and (c) Operating Costs.
- 2.28 "Net Loan Proceeds" will mean (i) the sum of (a) the proceeds of any loan made to the Company and the proceeds from refinancing any such loan, plus (b) any amount released from cash escrow accounts established under any loan to the Company, reduced by (ii) the sum of (w) any amounts required to fund the Company's capital expenditures that are permitted to be withheld from such amounts for such purpose under this Agreement, (x) any and all expenses incurred by the Company in connection with such loan or refinancing, (y) amounts used as permitted under this Agreement to repay other indebtedness of the Company, and (z) amounts thereof retained as reserves under this Agreement for disbursements by the Company reasonably expected by the Managers to occur.
- 2.29 "Non-Recourse Deductions" has the meaning set forth in Regulations Section 1.704-2(c). The amount of Non-Recourse Deductions for a Company Accounting Year equals the excess, if any, of the net increase, if any, in the amount of Company Minimum Gain during that fiscal year, over the aggregate amount of any Distributions during that fiscal year of proceeds of a Non-recourse Liability that are allocable to an increase in Company Minimum Gain, determined according to the provisions of Regulations Section 1.704-2(c).
- 2.30 "Non-recourse Liability" has the meaning set forth in Regulations Section 1.704-2 (b) (3).

2.31 "Percentage Interest" will mean seventy percent (70%) for BCI and thirty percent (30%) for Global.

Provided, however, that if either Member fails to make all or any portion of any contribution required by Section 8.4 with respect to any Aircraft in the percentages set out in Section 0, each Member's Invested Capital and Percentage Interest for each Aircraft will be accounted for separately, and the Percentage Interest with respect to any Aircraft as to which the contribution required by Section 8.4 has not been made in the percentages set out in Section 0 will be adjusted, with respect to such Aircraft, to the percentage derived by multiplying the Invested Capital contributed by BCI by .93334 and the Invested Capital contributed by Global by 1.2 and dividing each result by the total of such products and further adjusted by adding interest on the excess over the amount required in Section 8.4 contributed, at the rate of five percent (5%) above the prime rate of interest published in the Western Edition of The Wall Street Journal, or any successor publication, adjusted daily for any change in such prime rate of interest.

- 2.32 "Profit" or "Loss" will mean, for each Company Accounting Year, an amount equal to the Company's net taxable income or loss for such Accounting Year, determined in accordance with Code Section 703(a) (for this purpose, all items of income, gain, loss or deduction required to be stated separately pursuant to Code Section 703(a)(1) will be included in computing such taxable income or loss), with the following adjustments:
  - (a) Any income of the Company that is exempt from federal income tax and is not otherwise taken into account in computing Profit or Loss will be added to such taxable income or loss;
  - (b) In the event the agreed fair market value of any Company asset is adjusted pursuant to Regulations Section 1.704-1(b)(2)(iv)(f) or other pertinent sections of such Regulations, the amount of such adjustment will be taken into account as Gain or Loss on Disposition of such asset for purposes of computing Profit or Loss; and in lieu of the depreciation, amortization and other cost recovery deductions taken into account in computing such taxable income or loss, there will be taken into account depreciation, amortization or other cost recovery computed with reference to the value of Company property approved by the Managers (if different from its adjusted tax basis) pursuant to Regulations Section 1.704-1(b)(2)(iv)(g) for such Company Accounting Year; and
  - (c) Notwithstanding any other provisions, any items which are specially allocated pursuant to paragraphs 9.3, 9.4, 9.5 and 9.6 will not be taken into account in computing Profit or Loss.
  - 2.33 "Property" will mean the Aircraft, the Engines, and any component or

part thereof.

- 2.34 "Purchase Agreement" will mean an agreement between Jetran, as seller, and the Company as purchaser for acquisition of the Aircraft or beneficial interests in the Aircraft.
- 2.35 "Regulations" will mean United States Treasury Department or Internal Revenue Service Regulations promulgated under the Code.
- 2.36 "Tax Matters Partner" is defined in paragraph 11.3 (which references the Code).

# 4. OFFICES.

3.1 Principal Business Office. The principal business offices of the Company will be located at the addresses of BCI and Global. The Company may have such other offices, either within or outside Illinois, as the Members may designate or as the business of the Company may require from time to time.

# Page 5

3.2 Registered Office. The Company will maintain a registered office in Delaware as required by the Act. The address of the registered office may be changed from time to time by the Members.

#### 4. MEMBERS.

- 4.1 Initial Members. The initial members of the Company will be BCI and Global.
- 4.2 Additional Members. No Additional Members may be admitted to the Company without the prior written consent of all parties which are then Members.
  - 4.3 Transfer of Membership.
  - (a) General. No Member, without the prior written consent of all parties who are then Members may sell, assign, transfer, give, pledge, hypothecate or otherwise encumber a Membership Interest whether now owned or hereafter acquired, or contract to do any of those things, except as expressly permitted by this Agreement and in accordance with its terms. The Company, by its execution of this Agreement, agrees that it will not cause or permit the transfer of any Membership to be made unless the transfer is permitted by this Agreement and has been made in accordance with its terms. However, nothing contained in this paragraph is intended or is to be construed to prevent a Member from assigning its economic interest in the Company, provided that such assignment (i) will not relieve the assigning Member of its obligations under this Agreement, (ii) will be subject to all of the terms and conditions of this Agreement and (iii) will not entitle

the assignee or its successor in interest to participate in the business of the Company or to vote, for the election of Managers or any other purpose, it being the intent of the parties that, unless and until any such assignee is admitted as a Member in accordance with this Agreement, no such assignment will confer upon the assignee any of the rights or obligations of a Member hereunder other than any assigned interest in profits, losses and Distributions to which the assigning Member may be entitled under the terms and conditions of this Agreement.

- (b) Pledge to Secure Loans. Each Member may pledge its Membership Interest to secure any loan to such Member.
  - (c) Right of First Offer.
  - (i) Prior to offering to sell a Membership Interest to any person or entity not then a Member, the Member desiring to sell his Membership Interest will first give to the other Members notice of a right of first offer with respect to sale of such Membership Interest specifying the sale price per percentage of Percentage Interest at which such Member intends to sell his Membership Interest. The other Members will have a period of thirty (30) days after the effective date of such notice to purchase the offering Member's Membership Interest at the offered price. If the other Members do not purchase within such thirty (30) day period the offering Member's Membership Interest at the offered price multiplied by the offering Member's Percentage Interest, the offering Member will be free to sell his Membership Interest, provided that the sale price per Percentage Interest is equal to or in excess of the sale price per Percentage Interest offered to the other Members, and provided further that such sale is closed within two hundred forty (240) days after the giving by the offering Member to the other Members of notice of such intent to sell. Any other sale of such Membership Interest will require a reoffering pursuant to this paragraph.
  - (ii) If more than one Member wishes to acquire the offering Member's Membership Interest and such Members cannot agree as to the Percentage Interest to be acquired by each, the Membership Interest of the offering Member will be divided among the Members wishing to acquire such Membership Interest in proportion to the then-Percentage Interest of each.
- (d) Additional Option to Purchase. If any Membership is levied upon, sequestered, administered by a receiver or a trustee in bankruptcy, or is encumbered or proposed to be encumbered pursuant to Delaware statute, the affected Member will give the Company written notice of such occurrence and the Company will for a period of sixty days after receipt of such notice have the right to purchase all of such Member's interest in the Company for a purchase price equal to the book value (total assets less total liabilities appearing upon the Company's books of account as of the close of the latest financial period for which Company records are kept occurring immediately prior to such event) multiplied by that Member's Percentage

Interest by giving notice of exercise of such option to the person then having the right to receive Distributions from that interest. Payment by the Company for the Membership interest purchased pursuant to the option granted in this paragraph may be made in cash at the time of exercise of the option to purchase or at the election of the Company in equal semi-annual installments for a period of five years commencing six months after such date. If the Company elects to pay the purchase price in installments the purchaser will deliver to the holder of such interest a promissory note in the amount of the purchase price elected to be paid in installments, which promissory note will bear interest at the prime rate published in the Western Edition of The Wall Street Journal, adjusted from time to time for any change in such prime rate.

#### (e) Additional Transfer Restrictions.

- (i) No Member will be permitted to transfer any portion of its Company interest or take any other action which would cause the Company to be (y) treated as a "publicly traded partnership" within the meaning of Code Section 7704 or (z) classified as a corporation within the meaning of Code Section 7701(a);
- (ii) No Member will be permitted to transfer all or any portion of its Company interest or to take any other action (including, in the case of any Member which is a corporation, limited liability company or

#### Page 7

partnership or a partner or member of a partnership or limited liability company which is a Member, a transfer of any interest in such partnership, limited liability company or corporation or in the partners, members or shareholders thereof) which would result in a termination of the Company as a partnership within the meaning of Code Section 708(b)(1)(B) (a "Tax Termination"), without the approval of the non-transferring Members;

- (iii) Unless arrangements concerning withholding are reasonably approved by the non-transferring Members (if such withholding is required of the Company), no Member will be permitted to transfer all or any portion of its interest in the Company to any Person, unless such Person is a United States Person as defined in Code Section 7701(a)(30) and is not subject to withholding of any federal tax; and
- (iv) No Member will be permitted to transfer all or any portion of its Company interest if such transfer will (y) cause the assets of the Company to be deemed "Plan Assets" under ERISA or its accompanying regulations or the Code or (z) result in any "prohibited transaction" under ERISA or its accompanying regulations affecting the Company.

- 4.4 Meetings of Members.
- (a) General. It is not expected that regular meetings of the Members will be held, or that an annual meeting of the Members will be required unless required by the Act.
- (b) Special Meetings. Special meetings of the Members, for any purpose, unless otherwise prescribed by statute, may be called by any Member or Manager.
- (c) Place of Meeting. Any Member or Manager may designate any place in the continental United States, either within or outside Arizona or Illinois, as the place for any meeting of Members called by that Member or Manager. If no designation is made, the place of meeting will be the principal business office of the Company in Illinois.
- (d) Notice of Meeting. Written or printed notice stating the place, day and hour of the meeting, and, in case of a special meeting, the purposes for which the meeting is called, will be delivered not less than ten nor more than fifty (50) days before the date of the meeting, either personally or by mail, by or at the direction of the Manager calling the meeting, to each Member. If mailed, such notice will be deemed to be delivered as provided in paragraph 13 below.
- (e) Quorum. The holders of a majority of the Percentage Interests of Members, represented in person or by proxy, will constitute a quorum at a meeting of Members. If less than a majority is represented at a meeting, a majority of the Percentage Interests so represented may adjourn the meeting from

Page 8

time to time without further notice. At such adjourned meeting at which a quorum is present or represented, any business may be transacted which might have been transacted at the meeting as originally notified. The Members present at a duly organized meeting may continue to transact business until adjournment, notwithstanding the withdrawal of enough Members to leave less than a quorum. If a quorum is present, the affirmative vote of a majority of Percentage Interests of Members represented at the meeting and entitled to vote on the subject matter will be the act of the Members, unless the vote of a greater number is required by law or this Agreement.

- (f) Proxies. At all meetings of Members, a Member may vote by proxy executed in writing by the Member or the president of the Member or his duly authorized attorney-in-fact. Such proxy will be filed with a Manager before or at the time of the meeting. No proxy will be valid after eleven (11) months from the date of its execution, unless otherwise provided in the proxy.
- (g) Voting of Memberships. Each Member will be entitled to one (1) vote for each one percent (1%) of Percentage Interests held by that Member, on each matter submitted to a vote at a meeting of Members. Except as otherwise

specified herein, no Member not a Member will have a right to vote. In the election of Managers each Member will have the right to vote the number of votes represented by its Percentage Interest multiplied by the number of Managers to be appointed by that Member. Cumulative voting will not be allowed for the election of Managers or any other purpose.

- (h) Voting of Memberships by Certain Holders. Memberships standing in the name of a corporation may be voted by its president or such officer, agent or proxy as the bylaws of such corporation may prescribe or, in the absence of such provision, as the president of such corporation may determine. Memberships held by an administrator, executor, guardian or conservator may be voted by him, either in person or by proxy, without a transfer of such shares into his name. Memberships standing in the name of a trustee may be voted by him, either in person or by proxy. No person or entity will be entitled to vote Memberships held by him without a transfer of such Membership into his name in accordance with the terms of this Agreement. A Member whose interest is pledged will be entitled to vote such interests until the Membership has been transferred into the name or the pledgee or successor in interest as permitted by and in accordance with the terms of this Agreement.
- (i) Action by Consent of Members. Any action required to be taken at a meeting of Members, or any other action which may be taken at a meeting of Members, may be taken without a meeting if a counterpart, but not necessarily the same counterpart, of a consent in writing, setting forth the action so taken, is signed by the number of Members required for passage of a vote upon such action. Such consent will have the same force and effect as a unanimous vote of the Members, and may be stated as such in any articles or document filed with the Secretary of State of Delaware under the Act.

Page 9

#### 5. MANAGERS.

- 5.1 Number, Tenure and Qualifications. The initial number of Managers will be two, one of which will be appointed by BCI and the other of which will be appointed by Global. The Member appointing a Manager may remove that Manager at any time by giving written notice to the other Member. In the event of a vacancy the Member appointing the Manager the post of which is vacant will have the power to appoint a substitute or replacement Manager. Each Manager will hold office until his successor has been appointed and qualified. Managers need not be residents of Delaware or Members of the Company.
- 5.2 General Powers. Except as otherwise specified in the Act and as limited herein, the business and affairs of the Company will be managed by its Managers. Without limiting the foregoing, the Managers are authorized on behalf of the Company to make all decisions as to the improvement, sale, lease, or other disposition of the Property and other assets of the Company, as to the management of all or any part of the Company's business, as to the borrowing of

money and the granting of mortgages, and security interests in Company assets, as to the prepayment, refinancing, or extension of any loan to the Company, as to the compromise or release of any claims or debts of the Company, and as to the employment of persons, firms, or corporations for the operation and management of the Company's business. In order to implement its management powers, the Managers, and each of them, are authorized to execute and deliver (a) all sales contracts, assignments, bills of sale, leases, parts purchase and refurbishment contracts, management contracts and maintenance contracts covering or affecting the Property; (b) all checks, drafts and other orders for the payment of Company funds; (c) all promissory notes, mortgages, security agreements and other similar documents; and (d) all other instruments of any kind or character relating to the affairs of the Company whether like or unlike the foregoing. In addition, the Managers will be authorized to enter into agreements on behalf of the Company in such form as both Managers may approve, contracting with any Member or other persons or entities; whether or not affiliated with any Member, for management, disposition, refurbishment or other services. The compensation of a Member under any such agreement will be competitive with compensation payable under similar arms-length agreements in the area where the Member is located and the services are performed.

- 5.3 Limitations Upon Powers of Managers. Notwithstanding the foregoing, the Managers may not without the prior written consent of all of the Members where indicated and the Members where not indicated:
  - (a) Make any material change to the Company's business;
  - (b) Amend this Limited Liability Company Agreement or the Certificate of Formation of the Company (consent of all Members);
    - (c) Issue or sell any Membership Interest in the Company;
  - (d) Except as provided in a separate written agreement between the Company and a Member or a Manager, approved by the Members, set or revise the compensation of any Member or Manager or any affiliate of a Member or

## Page 10

Manager, whether acting in the capacity of a member or manager or whether providing services or supplies to the Company in the usual business of such Manager or Member;

- (e) The purchase or sale of any asset having a value in excess of one hundred thousand dollars (\$100,000.00);
- (f) Excepting Member Loans, borrowing in amounts greater than one hundred thousand dollars (\$100,000) in the aggregate;
- (g) Any change to the Company's accountants or method of accounting for book or tax purposes;

- (h) The withholding of any cash distribution to the Members in excess of current working capital needs and reasonable reserves for future expenditures.
- 5.4 Granting of Security Interests in the Property. Notwithstanding the foregoing, the Manager appointed by BCI may grant a security interest in all or any portion of the Property to secure any loan to BCI, the proceeds of which are used by BCI to make loans or capital contributions to the Company or to refinance loans used for such purpose.
- 5.5 Vacancies. Any Manager may resign at any time by giving written notice to the Members. Such resignation will take effect at the time specified therein; and unless otherwise specified therein, the acceptance of such resignation will not be necessary to make it effective.
- 5.6 Manner of Acting. The act of both Managers at any meeting or by consent will constitute an act of the Managers. In addition, the Managers may designate in writing a Manager or other person to act on behalf of the Managers for the purpose or purposes and subject to the limitations stated in such writing. If the Managers cannot agree as to any action to be taken by the Managers, such disagreement will be referred for decision to the presidents of the Members. If the presidents of the Members cannot reach a decision with respect to such action within ten (10) business days after referral of such matter to them, either Manager or Member may refer the matter for decision to the American Arbitration Association for arbitration by one arbitrator in accordance with Section 16.7.
- 5.8 Informal Action by Managers. Any action required or permitted to be taken at a meeting of the Managers may be taken without a meeting if a counterpart, but not necessarily the same counterpart, of a consent in writing, setting forth the action so taken, is signed by the requisite number of the Managers entitled to vote with respect to the subject matter thereof. Such consent will have the same force and effect as a unanimous vote of the Managers, and may be stated as such in any articles or documents filed with the Secretary of State of Delaware under the Act.

Page 11

# 6. COSTS AND EXPENSES.

All costs and expenses of the Company will be paid from its funds, including but not limited to (a) costs of forming and organizing the Company including legal, accounting and consulting; (b) the costs of acquiring the Property and other properties and assets of the Company; (c) the costs of reconditioning and refurbishing any of the Property; (d) the costs of sale, lease, and other disposition of the Property; (e) legal fees and accounting fees; (f) reimbursement of costs and expenses incurred by the Managers and Members in the performance of their duties for the Company; and (g) every other

cost and expense incurred in the Company's business whether like or unlike the foregoing. Provided, however, that without the prior written consent of both Members, costs of reconditioning and refurbishing the Aircraft will not exceed the amounts set forth in the Reconditioning Cost Schedule attached hereto.

# 7. INDEMNIFICATION OF MANAGERS.

Each Manager and each person who serves at his request, whether or not then in office, and his personal representatives, will be indemnified by the Company against all costs and expenses actually and necessarily incurred by him in connection with the defense of any action, suit or proceeding in which he may be involved or to which he may be made a party by reason of his being or having been a Manager, to the fullest extent permitted by the Act, as is now or hereafter amended. Without limiting the foregoing, the Company will indemnify any Manager of the Company made a party to any legal proceeding in his capacity as a Manager if the indemnitee (a) conducted himself in good faith; (b) reasonably believed that his conduct in his official capacity with the Company was in the best interest of the Company, and in the case of any other conduct that such conduct was at least not opposed to the Company's best interests; and (c) in the case of any criminal proceeding, he had no reasonable cause to believe that his conduct was unlawful, provided that the termination of any such legal proceeding by judgment, order, settlement, or conviction, or upon a plea of nolo contendere or its equivalent will not itself be determinative if the individual did not meet the standards of conduct set forth herein. The Company may pay for or reimburse the reasonable expenses incurred by a Manager in advance of the filed disposition of the proceeding if the Manager furnishes the Company with a written affirmation of this good faith belief that he has met the standard of conduct set forth above, and he furnishes the Company with a written undertaking executed personally or on his behalf to repay the amounts advanced if it is finally determined that he did not meet the standard of conduct set forth above. In addition to the foregoing, the Company will pay or reimburse expenses incurred by a Manager in connection with his appearance as a witness in any legal proceedings during the period of time for which he has not been named as a defendant or respondent in such legal proceeding.

#### 8. CAPITAL CONTRIBUTIONS; LOANS BY MEMBERS.

8.1 General. Except as otherwise stated in this paragraph below, or as agreed between the contributing Member and both Managers, no Member will be required or permitted to make a capital contribution or loan to the Company or will be otherwise responsible for the debts or obligations of the Company.

Page 12

8.2 Form of Contributions. Capital contributions to the Company may be made in the form of cash or real or personal property. The Company may assume any obligations of the contributing Member with respect to property contributed in writing at the time of acceptance of such contribution by the Managers.

- 8.3 Basis of Non-Cash Contributions. If real or personal property is contributed to the Company the value of such contribution will be the cost basis of such property or such other amount as may be agreed in writing by the contributing Member and the Managers. The capital account of the Member will be credited with the amount of the contribution net of liabilities assumed by the Company at the time of the acceptance of the contribution by the Managers.
- 8.4 Contributions of Amounts Required to be Paid in Accordance with the Purchase Agreement.
  - (a) Deposit. Global has paid fifty percent (50%) of the initial one million dollar (\$1,000,000.00) deposit required under Section 1.2(a)(i) of the purchase agreement. BCI will reimburse Global two hundred fifty thousand dollars (\$250,000) of such deposit and reimburse Jetran for five hundred thousand dollars (\$500,000) of such deposit.
  - (b) Additional Payments. All additional payments required to be made under the Purchase Agreement will be made seventy five percent (75%) by BCI and twenty five percent (25%) by Global, as follows:
    - (i) Upon execution hereof, the Members will together place on deposit with Wells Fargo Bank Northwest, National Association three million five hundred thousand dollars (\$3,500,000), with instructions to pay to the seller-agent in accordance with Section 1.2(c) of the Purchase Agreement promptly upon receipt of the required deposit from both the Members.
    - (ii) Not later than two (2) business days prior to the date set for closing of each Aircraft under the Purchase Agreement, each of the Members will together place on deposit with Wells Fargo Bank Northwest, National Association, the purchase price required under Section 1.2(d) of the Purchase Agreement to be paid at closing, with instructions to forward such purchase price to the seller-agent in accordance with Section 1.2(d) of the Purchase Agreement.
- 8.5 Additional Funds Required for the Company's Business. Any additional funds required to be advanced for the Company's business may be contributed by the Members. 8.6 Member Loans. Any additional amounts required for the Company's business and not contributed by the Members may be loaned by either Member to the Company at the request of the Managers. Such loans are referred to as "Member Loans." Member Loans will bear interest at the rate of five percent (5%) above the prime rate of interest published in the Western Edition of The Wall Street Journal, or any successor publication, adjusted daily for any change in such prime rate of interest and will be repaid prior to the payment of any distribution to the Members pursuant to Section 10.

#### 9. ALLOCATIONS.

Allocations of a Member's interest in capital of the Company, its share of the Profits and Losses, its share of the Distributions of the Company, its Capital Account, and its other rights and obligations under this Agreement will be allocated in accordance with its Percentage Interest, except as otherwise provided in this paragraph 9.

9.1 Establishment and Maintenance of Capital Accounts. The Manager will establish and maintain a single Capital Account for each Member which reflects each Member's Capital Contributions to the Company. Each Capital Account will also reflect the allocations and Distributions made pursuant to paragraphs 9 and 10 and otherwise be adjusted in accordance with Code Section 704 and the principles set forth in Regulations Sections 1.704-1(b) and 1.704-2. In applying such principles, any expenditures of the Company described in Code Section 705(a)(2)(B) or treated as Code Section 705(a)(2)(B) expenditures pursuant to Regulations Section 1.704-1(b)(2)(iv)(i) will be allocated among the Members in proportion to their respective Percentage Interests. The Members intend that the Company be treated as a partnership for tax purposes.

#### 9.2 Profit and Loss Allocations.

- (a) Operating Losses. For each Company Accounting Year from the Agreement Date until the termination of the Company, Loss from Company operations will be allocated among the Members in the following order of priority (for purposes of determining Capital Accounts under this paragraph 9.2(a) and 9.2(b): (y) Capital Accounts will first be reduced by Distributions with respect to such Company Accounting Year and (z) a Member's Capital Account balance will be deemed to be increased by such Member's share of Company Minimum Gain and Member Non-recourse Debt Minimum Gain determined as of the end of such Company Accounting Year):
  - (i) First, among the Members pro rata as necessary to cause each Member's Capital Account balance, for the Company Accounting Year, to equal zero; and
  - (ii) Then, among the Members in proportion to their then respective Percentage Interests.
- (b) Operating Profits. For each Company Accounting Year, Profits from the Company's operations will be allocated among the Members in proportion to their then respective Percentage Interests.

Page 14

9.3 Allocations of Gain or Loss on Disposition. For purposes of determining

Capital Account balances under this paragraph 9.3, Gain or Loss on Disposition will be allocated prior to reducing Capital Accounts by the Distributions of Capital Receipts from the Disposition.

- (a) Gain on Disposition will be allocated among the Members in the following order of priority:
  - (i) First, to the Members in proportion to, and to the extent of, any deficit balances in their respective Capital Accounts until all such Capital Accounts have been restored to zero;
  - (ii) Then, among the Members in proportion to their then respective Percentage Interests.
- (b) Loss on Disposition will be allocated among the Members in the following order of priority:
  - (i) First, among the Members pro rata as necessary to cause the Capital Account balance of each Member to equal zero; and
  - (ii) Then, among the Members in proportion to their then respective Percentage Interests.
  - (c) Rules of Construction.
  - (i) For purposes of applying paragraph 9.3 as a result of a Disposition occurring with respect to part (but less than all) of any capital asset of the Company, a Member's Capital Account balance will be deemed to be increased by such Member's share of Company Minimum Gain and Member Non-Recourse Debt Minimum Gain remaining after such Disposition as determined under the Regulations under Section 704(b).
  - (ii) If there is insufficient Profit, Loss or Gain or Loss on Disposition to allocate to the Members pursuant to any subparagraph of paragraph 9.2 or 9.3 to cause every Member's Capital Account balance to equal the entire Capital Account balance described in such subparagraph with respect to such Member, the Profit, Loss or Gain or Loss on Disposition available to be allocated among the Members pursuant to said subparagraph will be allocated in proportion to the amounts thereof that would have been allocated to each Member pursuant to such subparagraph if there had been sufficient amounts thereof to fully satisfy the requirements of such subparagraph with respect to every Member.
  - (iii) Except as is otherwise provided in this paragraph 9, an allocation of Company taxable income or taxable loss to a Member will be treated as an allocation to such Member of the same share of each

item of income, gain, loss and deduction that has been taken into account in computing such taxable income or taxable loss.

- 9.4 Minimum Gain Chargeback and Qualified Income Offset.
- (a) No Impermissible Deficits. Notwithstanding any other provision of this Agreement, taxable loss (or items of deduction) will not be allocated to a Member to the extent that the Member has or would have, as a result of such allocations, an Adjusted Capital Account Deficit. Any taxable loss (or items of deduction) which otherwise would be allocated to a Member, but which cannot be allocated to such Member because of the application of the immediately preceding sentence, will instead be allocated to the other Members.
- (b) Qualified Income Offset. In order to comply with the "qualified income offset" requirement of the Regulations under Code Section 704(b), and notwithstanding any other provision of this Agreement to the contrary except paragraph 9.4(c) below, in the event a Member for any reason (whether or not expected) has an Adjusted Capital Account Deficit, items of Profits and Gain on Disposition (consisting of a pro rata portion of each item of income comprising the Company's Profits and Gain on Disposition, including both gross income and gain for the taxable year) will be allocated to such Member in an amount and manner sufficient to eliminate as quickly as possible the Adjusted Capital Account Deficit.
- (c) Minimum Gain Chargeback. In order to comply with the "minimum gain chargeback" requirements of Regulations Sections 1.704-2(f)(1) and 1.704-2(i)(4), and notwithstanding any other provision of this Agreement to the contrary, in the event there is a net decrease in a Member's share of Company Minimum Gain and/or Member Non-Recourse Debt Minimum Gain during a Company taxable year, such Member will be allocated items of income and gain for that year (and if necessary, other years) as required by and in accordance with Regulations Sections 1.704-2(f)(1) and 1.704-2(i)(4) before any other allocation is made.

# 9.5 Other Tax Allocation Provisions.

(a) Income Characterization. For purposes of determining the character (as ordinary income or capital gain) of any Gain on Disposition allocated to the Members pursuant to paragraph 9.3 or 9.4, such portion of the taxable income of the Company allocated pursuant to paragraph 9.3 which is treated as ordinary income attributable to the recapture of depreciation will, to the extent possible, be allocated among the Members in the proportion which (i) the amount of depreciation previously allocated to each Member bears to (ii) the total of such depreciation allocated to all Members. This paragraph 9.5(a) will not alter the amount of allocations among the Members pursuant to paragraph 9.3 but merely the character of income so allocated.

- (b) Change in Percentage Interests. Notwithstanding the foregoing, in the event any Member's Percentage Interest changes during a fiscal year for any reason, including the transfer of any interest in the Company or an adjustment of the Members' Percentage Interests hereunder, the allocations of taxable income or loss under this paragraph 9, and Distributions, will be adjusted as necessary to reflect the varying interests of the Members during such year using an interim closing of the books method as of the date of such change or such other method as is approved by the Manager.
- (c) Mandatory Allocations -- Section 704(c) and Member Non-Recourse Debt.
  - (i) Notwithstanding the foregoing, (i) in the event Code Section 704(c) or Code Section 704(c) principles applicable under Regulations Section 1.704-1(b)(2) (iv) require allocations of income or loss of the Company in a manner different than that set forth above, the provisions of Section 704(c) and the regulations thereunder will control such allocations among the Members; and (ii) all tax deductions and taxable losses of the Company that, pursuant to Regulations Section 1.704-2(i), are attributable to a Member Non-Recourse Debt for which a Member (or a Person related to such Member under Treasury Regulations Section 1.7524(b)) bears the economic risk of loss (within the meaning of Regulations Section 1.752-2) will be allocated to such Member as required by Regulations Section 1.704-2(c).
  - (ii) Any item of income, gain, loss and deduction with respect to any property (other than cash) that has been contributed by a Member to the capital of the Company or which has been revalued for Capital Account purposes pursuant to Regulations Section 1.704-1(b)(2)(iv) and which is required or permitted to be allocated to such Member for income tax purposes under Section 704(c) of the Code so as to take into account the variation between the tax basis of such property and its fair market value at the time of its contribution will be allocated solely for income tax purposes in the manner so required or permitted under Code Section 704(c) using the "traditional method" described in Regulation Section 1.704-3(b) (or any successor Regulation), such allocations to be made as will be reasonably approved by the Manager; provided, however, that curative allocations consisting of the special allocation of gain or loss upon the sale or other disposition of the contributed property will be made in accordance with Regulations Section 1.704-3(c) to the extent necessary to eliminate any disparity, to the extent possible, between the Members' book and tax Capital Accounts attributable to such property; and further provided, however, that any other method allowable under applicable Regulations may be used in connection with any contribution of property or following any revaluation pursuant to paragraph 9.9 as will be determined by the

Manager. Any elections or other decisions relating to allocations under this paragraph 9.5(c)(ii) will be made in any manner that the Manager reasonably determines to reflect the purpose and intention of this Agreement. Allocations under this paragraph 9.5(c)(ii) are solely for purposes of federal, state and local taxes and will not affect, or in

#### Page 17

any way be taken into account in computing, any Member's Capital Account or share of Profit, Loss, Gain or Loss on Disposition or other items or Distributions under any provision of this Agreement.

(d) Guarantee of Company Indebtedness. Except for arrangements expressly described in this Agreement (including loans described in paragraph 8), no Member other than the Managers will enter into (or permit any Person related to the Member to enter into) any arrangement with respect to any liability of the Company that would result in such Member (or a Person related to such Member under Regulations Section 1.752-4(b)) bearing the economic risk of loss (within the meaning of Regulations Section 1.752-2) with respect to such liability unless such arrangement has been approved by the Manager. This paragraph 9.5(d) will not prohibit a Member from making a loan described in paragraph 8. To the extent a Member is permitted to guarantee the repayment of any Company indebtedness under this Agreement, each of the other Members will be afforded the opportunity to quarantee such Member's pro rata share of such indebtedness, determined in accordance with the Members' respective Percentage Interest. Notwithstanding the foregoing, but subject to the following provisions of this paragraph 9.5(d), the Manager will have the right to cause the Company, or subsidiary of the Company, as the case may be, to borrow funds from third party lenders, provided the Manager will use its reasonable efforts to cause any such loans to be made to the Company, and/or subsidiary of the Company, as the case may be, on a basis that is fully non-recourse to the Members or on a basis that is recourse only to the Members who have so approved such recourse liability. In the event that a loan is to be made to the Company, or subsidiary of the Company, as the case may be, and such loan is to be guaranteed by any Members (which will occur only upon the approval of such Members), then, as between such approving Members, such liability will be shared in proportion to their respective Percentage Interest (and, if joint and several liability of such Members will be required by the lender under a Company borrowing or a borrowing of a subsidiary of the Company, as the case may be, each such approving Member will make contributions to the Company when requested by the Manager to do so, which amounts will immediately be distributed to such other approving Members, as necessary for such approving Members to bear the economic risk of loss with respect to any such borrowing in proportion to their then respective Percentage Interest). The Managers will make such request at the direction of any Member.

(e) References to Regulations. Any reference in this Agreement to a provision of final, proposed and/or temporary Regulations will, in the event such provision is modified or renumbered, be deemed to refer to the successor provision as so modified or renumbered, but only to the extent such successor provision applies to the Company under the effective date rules applicable to such successor provision or the Managers otherwise so elect under applicable elections contained in such Regulations.

- 9.6 Intent of Allocations. The parties intend that the foregoing tax allocation provisions of this paragraph 9 will produce final Capital Account balances of the Members that will permit liquidating Distributions that are made in accordance with final Capital Account balances under paragraph 10.2(c) hereof to be made (after unpaid loans and interest thereon, including those owed to Members have been paid) in a manner identical to the order of priorities set forth in paragraph 10.1. To the extent that the tax allocation provisions of this paragraph 9 would fail to produce such final Capital Account balances, (i) such provisions will be amended by the Managers if and to the extent necessary to produce such result and (ii) taxable income and taxable loss of the Company for prior open years (or items of gross income and deduction of the Company for such years) will be reallocated by the Managers among the Members to the extent it is not possible to achieve such result with allocations of items of income (including gross income) and deduction for the current year and future years, as approved by the Managers. This paragraph 9.6 will control notwithstanding any reallocation or adjustment of taxable income, taxable loss, or items thereof by the Internal Revenue Service or any other taxing authority.
- 9.7 Basis Elections. In the event of a transfer of all or any part of a Member's interest in the Company, the Company will elect to adjust the basis of the Company's assets under Code Section 754 if approved by the Managers. The transferor or transferee of a Company interest will pay all costs of preparing and filing all instruments or documents necessary to effectuate such election if made.
- 9.8 General Allocation Rules. All Profit and Loss of the Company and Gain or Loss on Disposition will be allocated with respect to each Company Accounting Year (or part thereof) as of the end of, and within ninety (90) days after the end of, such year, or as soon thereafter as is practically possible. All Profit and Loss and Gain or Loss on Disposition will be allocated to the Members shown on the records of the Company to have been Members as of the last day of the Company Accounting Year for which such allocation is to be made, except that, if a Member sells or exchanges its interest in the Company or otherwise is admitted as a substituted Member, the Profit or Loss and Gain or Loss on Disposition will be allocated between the transferor and the transferee by taking into account their varying interests during the Company Accounting Year in accordance with Code Section 706(d), using the interim closing of the books method or such other method as will be approved by the Managers.

- 9.9 Sharing of Company Non-Recourse Debt. Throughout the term of the Company, the non-recourse debt of the Company (other than Member Non-Recourse Debt) will be allocated for tax purposes among the Members in accordance with their then respective Percentage Interests. To the extent that any Member's share of such non-recourse debt as so specified exceeds the amounts referred to in Regulations Sections 1.752-3(a)(1) and (2), it is intended that the foregoing shares will be viewed and treated as reasonably consistent with allocations (which have substantial economic effect) of some significant item of partnership income or gain within the meaning of Regulations Section 1.752-3(a)(3).
- 9.10 Adjustment of Gross Asset Value. Gross Asset Value, with respect to any asset, is the adjusted basis for federal income tax purposes of that asset, except as follows:

- (a) The initial Gross Asset Value of any asset contributed (or deemed contributed under Regulations Section 1.708-1(b)(l)(iv)) by a Member to the Company will be the fair market value of the asset on the date of the contribution, as reasonably approved by the Managers.
- (b) The Gross Asset Values of all Company assets will be adjusted to equal the respective fair market values of the assets, as reasonably approved by the Manager:
  - (i) If the Managers approve that an adjustment is necessary or appropriate to reflect the relative economic interests of the Members in the Company, as a result of (y) the acquisition of an additional interest in the Company by any new or existing Member in exchange for more than a de minimis capital contribution; or (z) the distribution by the Company to a Member of more than a de minimis amount of Company property as consideration for an interest in the Company; and
  - (ii) As of the liquidation of the Company within the meaning of Regulations Section 1.704-1(b)(2)(ii)(g).
- (c) The Gross Asset Value of any Company asset distributed to any Member will be the gross fair market value of the asset on the date of distribution as reasonably approved by the Managers.
- (d) The Gross Asset Values of Company assets will be increased or decreased to reflect any adjustment to the adjusted basis of the assets under Code Section 734(b) or 743(b), but only to the extent that the adjustment is taken into account in determining Capital Accounts under Regulations Section 1.704-1(b)(2)(iv)(m), provided that Gross Asset Values will not be adjusted under this paragraph 9.10(d) to the extent that the Managers approve that an adjustment under paragraph 9.10(b) is necessary or

appropriate in connection with a transaction that would otherwise result in an adjustment under this paragraph 9.10(d).

- (e) After the Gross Asset Value of any asset has been determined or adjusted under paragraph  $9.10\,(a)$ ,  $9.10\,(b)$  or  $9.10\,(d)$ , Gross Asset Value will be adjusted by the depreciation taken into account with respect to the asset for purposes of computing Profits or Losses.
- 9.11 Section 1031 Exchanges. If any Member desires to defer income recognition in connection with the sale of an asset by the Company under Code Section 1031, and the Managers so approve, but one or more other Members do not desire to do so with respect to their indirect economic interest in such asset, such other Members agree to cooperate with such Member and the Managers in restructuring the sale of such asset as a like-kind exchange under Code Section 1031 with respect to the indirect economic interest in such asset of such Member, to the extent possible, provided, however, that amount of the cash received by such other Members in connection with such restructured sale is the same as it would have been (and is received not later than it would have been received) if the sale were not restructured as a like-kind exchange.

Page 20

## 10. MEMBER LOAN REPAYMENTS AND DISTRIBUTIONS.

- 10.1 Net Available Cash, Net Loan Proceeds and Capital Receipts. The Managers will, (y) at the end of each quarter, determine the amount of Net Available Cash, if any, and apply or distribute Net Available Cash as set forth below within thirty (30) days after the end of each such quarter, and, (z) upon the occurrence of any event giving rise to Net Loan Proceeds or Capital Receipts, determine the amount of Net Loan Proceeds and Capital Receipts, if any, and apply or distribute (as applicable) Net Loan Proceeds and Capital Receipts as set forth below within fifteen (15) days after receipt thereof by the Company, in each case, subject to the terms of paragraph 10.3 hereof, in the following order of priority.
  - (a) First, payments will be made to the Members in repayment of any Member Loans (loans which have been outstanding the longest will be repaid first, and if two or more Members have loans which have been outstanding for equal periods, repayment of Member Loans will be made pro rata, in proportion to the lending Members' then respective loan balances, with payments first repaying accrued but unpaid interest and then repaying principal);
  - (b) Next, Distributions will be made to the Member, pro rata in proportion to the amount by which the Invested Capital of BCI exceeds 75% of the total Invested Capital or by which the Invested Capital of Global exceeds 25% of the total Invested Capital

- (c) Next, the balance will be distributed to the Members, pro rata, in proportion to their resultant Capital Accounts.
- 10.2 Proceeds and Distributions in Liquidation. The proceeds received by the Company in connection with the liquidation and winding up of the Company will be applied by the Liquidator in the following order of priority:
  - (a) First, to the payment of the expenses incurred in dissolution and termination;
  - (b) Next, to the payment of creditors of the Company (including making the repayments of any Member loans as set forth in paragraph 10.1(a)) except secured creditors whose obligations will be assumed or otherwise transferred on a liquidation of the Company property or assets; and
  - (c) The balance, if any, will be distributed to the Members to the extent of and in proportion to the positive balances of their Capital Accounts after Capital Accounts have been adjusted for the allocation of Profit and Loss (and items thereof) and Gain or Loss on Disposition and any adjustment required by Section 0 for the Company Accounting Year during which such liquidation occurs and all prior periods.

- 10.3 General Distribution Rules. The timing and amount of all Distributions will be in accordance with paragraphs 10.1, 10.2, 12.3 and 12.4. All Distributions of cash will be made to the Members shown on the records of the Company to have been Members on the date of the distribution. All Distributions, upon request by a Member, will be made by wire transfer in immediately available funds to such Member's account specified in such request. Distributions of Net Available Cash, Net Loan Proceeds and Capital Receipts made to a Member with respect to any Company Accounting Year will be deemed to be advances on account of such Member's share of the distributable amounts thereof. For purposes of this Agreement, the term "distributable" with respect to such Distributions will mean the amount of such Distributions as finally determined by the Manager for the Company Accounting Year in respect of which they were made. Any over-distribution thereof to any Member in respect of a Company Accounting Year will be repaid by such Member to the Company and distributed to the Member(s) which has received an under-distribution thirty (30) days after the end of such Company Accounting Year.
- 10.4 Source of Distributions. Each Member will look solely to the assets of the Company for the return of its Capital Contributions and its share of Distributions of the Company and will have no recourse upon dissolution or otherwise against the Company, the Managers, the Liquidator or any other Member. No holder of an interest in the Company will have any right to receive any Distributions except as provided in this Agreement or any right to demand or receive property other than cash upon dissolution and termination of the

Company.

#### 11. BOOKS AND RECORDS.

- 11.1 Books of Account. At Company expense, the Managers will cause to be kept at the office of the Company referred to in paragraph 3.1 accurate books of account, in which will be entered fully and accurately each and every transaction of the Company. The books will be kept in accordance with the Company's method of reporting for federal income tax purposes. Tax accounting elections, including methods of depreciation and deduction or capitalization of interest, taxes and insurance premiums during a construction period, if any, will be made as the Managers will approve as being in the best interests of the Company and its Members. At Company expense, the Managers will cause annual financial statements of the Company's operations to be prepared; any such financial statements to be prepared in accordance with federal income tax principles, consistently applied (except insofar as any change in the application thereof is disclosed in such financial statements).
- 11.2 Reports. The books of account will be closed promptly after the end of each calendar year. Promptly thereafter, the Managers will make a written report to each Member which will include a statement of income and expenses for the year, a statement of each Member's capital and undistributed profits accounts and such additional statements with respect to the status of the Company property and the distribution of Company funds as are considered necessary by the Managers to advise all Members properly about their investment in the Company. Such report may consist in part of a copy of part or all of the Company's United States income tax return. Prior to March 31 of each year each Member will also be provided with an information letter with respect to its distributive shares of income, gains, deductions and credits for income tax reporting purposes for the previous year. Such financial statements and the

Page 22

information contained therein will be deemed conclusive and binding upon such Member unless written objection is lodged with the Managers within ninety (90) days after the giving of such reports to such Member.

11.3 Tax Matters Partner. BCI is hereby designated as the "Tax Matters Partner", as such term is defined in Section 6231(a)(7) of the Code, and it will serve as such at Company expense with all powers granted to a tax matters partner under the Code. BCI will use its best efforts to cause the Company's accountants to prepare and file on a timely basis, with due regard to extensions (such extensions to be applied for or elected, as approved by the Managers), all tax and information returns which the Company may be required to file. No tax or information return will be filed unless approved by the Managers. Each Member will give prompt notice to each other Member of any and all notices it receives from the Internal Revenue Service concerning the Company, including any notice of audit, any notice of action with respect to a revenue agent's report, any

notice of a 30-day appeal letter and any notice of a deficiency in tax concerning the Company's federal income tax return. The Tax Matters Partner will furnish each Member with status reports regarding any negotiation between the Internal Revenue Service and the Company. The Tax Matters Partner will cause the Company's accountants to prepare and deliver, at Company expense, to each Member on a timely basis an information reporting return (K-1) reflecting each Member's distributive share of all income, gain, loss, deductions, allowances or credits of the Company for each Company Accounting Year. The Tax Matters Partner will deliver to all Members copies of all tax and financial reports when received from its accountant.

#### 12. DISSOLUTION AND TERMINATION.

- 12.1 Continued Existence. If there are two or more remaining Members, if an event effecting a dissolution in accordance with this Agreement, the Certificate of Formation of the Company or the Act, the remaining Members may elect in writing within ninety (90) days after the occurrence of such event to continue the existence of the Company, in which case, without further action, the Company will continue to exist. Absent such election, the Company will dissolve in the ninety first (91st) day following such event.
- 12.2 Final Accounting. In case of the dissolution of the Company, a proper accounting will be made as provided in paragraph 11.1 from the date of the last previous accounting to the date of dissolution.
- 12.3 Liquidation. Upon the dissolution of the Company, the Managers or, in the case of the withdrawal, resignation, death, incompetency, insolvency or bankruptcy of any Manager, the survivors, or in the case of the withdrawal, resignation, death, incompetency, insolvency or bankruptcy of all Managers, some person selected by the holders of a majority of the Percentage Interests will act as liquidator to wind up the Company. The liquidator will have full power and authority to sell, assign and encumber any or all of the Company's assets needed to wind up and liquidate the affairs of the Company in an orderly and businesslike manner. All proceeds from liquidation will be distributed in accordance with the Act as then in effect in the order of priority specified in paragraph 10.2 above. The liquidator will use reasonable efforts to distribute the proceeds from a liquidation in the same calendar year in which the sale of Company assets occur.

Page 23

12.4 Timing Requirements of Treasury Regulations. Notwithstanding anything in this paragraph 12 to the contrary, in the event the Company is "liquidated" within the meaning of Regulations Section 1.704-1(b)(2)(ii)(g), Distributions will be made to the Members who have positive Capital Account balances pursuant to paragraph 10.2 in a manner that complies with Regulations Section 1.704-1(b)(2)(ii)(b)(2). However, a liquidation occurring as a result of a Tax Termination will not require an actual distribution of Company assets, but will instead be treated as a constructive liquidation and reformation in the manner

in Regulations Section 1.708-1(b)(1)(iv).

12.5 Articles of Dissolution. Upon the completion of the distribution of Company assets as provided in paragraph 10.2, the Company will be dissolved, and the person acting as liquidator (or the Members, if necessary) will cause the filing of Articles of Dissolution with the Secretary of State and will take such other actions as may be necessary to dissolve the Company.

## 13. NOTICES

- 13.1 Method for Notice. All notices hereunder will be sent by certified or registered mail addressed as set forth herein, or if not set forth herein appearing in the records of the Company (except that any Party may from time to time give notice changing his address for that purpose) and will be effective on the date set forth on the receipt of registered or certified mail.
- 13.2 Computation of Time. In computing any period of time under this Agreement, the day of the act, event or default from which the designated period of time begins to run will not be included. The last day of the period so computed will be included, unless it is a Saturday, Sunday or legal holiday, in which event the period will run until the end of the next day which is not a Saturday, Sunday or legal holiday.

#### 14. TERM OF AGREEMENT.

Unless terminated sooner by unanimous agreement in writing of the Company and the Members, this Agreement will terminate upon the occurrence of any of the following events: (a) cessation of the Company's business; (b) sale of all non cash assets of the Company; (c) agreement of all of the Members; or (d) the happening of any event of termination specified in the Certificate of Formation of the Company.

#### 15. AMENDMENT AND MODIFICATION.

This Agreement may not be amended or modified except by mutual agreement in writing executed by the Company and signed by all Members.

#### 16. MISCELLANEOUS.

16.1 Financing Restriction (Tax-Exempt Investors). The Company will not borrow money from any source in connection with the acquisition or improvement of Company property or otherwise unless it first obtains an opinion of counsel that it is more likely than not that such borrowing will not result in any asset of the Company constituting "debt-financed property," as defined under Section 514(b) of the Code, with respect to any Member that is a "Qualified Organization," as defined under Section 514(c)(9)(C) of the Code.

- 16.2 Accountants. The Company's accountant will be selected by the Managers.
- 16.3 Status as Partnership for Tax Purposes. Any provision hereof to the contrary notwithstanding, solely for United States federal income tax purposes, each of the Members hereby confirms that the Company will be subject to all provisions of Subchapter K of Chapter I of Subtitle A of the Code; provided, however, the filing of U.S. Partnership Returns of Income will not be construed to extend the purposes of the Company or expand the obligations or liabilities of the Members.

## 16.4 Tax Withholding.

- (a) The Managers are authorized and directed to cause the Company to withhold from or pay on behalf of any Member the amount of federal, state, local or foreign taxes that the Managers, after consultation with such Member, reasonably believe the Company is required to withhold or pay with respect to any amount distributable or allocable to such Member pursuant to this Agreement, including, without limitation, any taxes required to be paid by the Company pursuant to Code Sections 1441, 1442, 1445 or 1446 and any taxes imposed by any state or other taxing jurisdiction on the Company as an entity. Without limiting the foregoing, the Managers will cause the Company to withhold (and remit to the appropriate governmental authority), from amounts otherwise distributable to a Member, any taxes that such Member notifies the Managers in writing should be withheld, which notice will be given by any Member who becomes aware of any withholding obligation to which it is subject and will specifically set forth, inter alia, the rate at which tax should be withheld and the name and address to which any amounts withheld should be remitted.
- (b) If the Company is required to withhold and pay over to taxing authorities amounts on behalf of a Member exceeding available amounts then remaining to be distributed to such Member, such payment by the Company will constitute a loan to such Member that is repayable by the Member on demand, together with interest at the applicable federal rate determined from time to time under Code Section 7872(f)(2) or the maximum rate permitted under applicable law, whichever is less, calculated upon the outstanding principal balance of such loan as of the first day of each month. Any such loan will be repaid to the Company, in whole or in part, as determined by the Managers in their sole discretion, either (i) out of any Distributions from the Company which the Member is (or becomes) entitled to receive, or (ii) by the Member in cash upon demand by the Member (said Member bearing all of the Company's costs of collection, including reasonable attorneys' fees, if payment is not remitted promptly by the Member after such a demand for payment).
- (c) Each Member agrees to cooperate fully with all efforts of the Company to comply with its tax withholding and information reporting obligations and agrees to provide the Company with such information as the

Managers may reasonably request from time to time in connection with such obligations.

- 16.5 Interest For Services. The Percentage Interest of any Member in excess of such Member's percentage of the capital contributions made by all Members will be deemed to be a profits interest received in exchange for services rendered or to be rendered to or for the benefit of the Company (such portion of any Member's Percentage Interest having no currently predictable Distributions).
- 16.6 Payment of Taxes Exceeding Distributions. Each Member understands that taxable income and gain allocated to such Member by the Company under this Agreement and the tax on the portion thereof allocated to such Member hereunder for any year may exceed the cash Distributions from the Company to such Member and such Member may have to look to sources other than Distributions from the Company to pay such taxes.
- 16.7 Arbitration. The parties to this Agreement will submit all controversies, claims and matters of difference related to this Agreement and all actions to be taken in accordance therewith. Arbitration will be conducted in the locale of one of the Company's business offices, according to the rules and practices of The American Arbitration Association from time to time in force. This submission and agreement to arbitrate will be specifically enforceable. Without limiting the generality of the foregoing, the following will be considered controversies for this purpose: (a) all questions relating to the construction of this Agreement, or the breach of any obligation, warranty or condition hereunder, (b) all questions related to any actions to be taken in accordance with this Agreement, (c) failure of any party to deny or reject a claim or demand of any other party to this Agreement, (d) all questions relating to the matters governed by this Agreement and (e) all guestions as to whether the right to arbitrate any questions exists. Arbitration may proceed in the absence of any party if notice of the proceedings has been given to such party in accordance with the rules of the American Arbitration Association. The parties agree to abide by all awards rendered in such proceedings. Such awards will be conclusive and binding on all parties to the extent and in the manner provided by applicable law. All awards may be filed with the clerk of one or more court, state or federal, having jurisdiction over the party against whom such an award is rendered or its property, as a basis of judgment and of the issuance of execution for its collection. No party will be considered in default under this Agreement during the pendency of arbitration proceedings relating to such default.
- 16.8 Attorneys' Fees and Costs. If any party to this Agreement brings a lawsuit or arbitration to enforce or interpret any of the provisions hereof or any action to be taken in connection herewith, any party against whom judgment is awarded will be liable for all costs and expenses including reasonable attorneys' fees incurred in the preparation, prosecution or appeal thereof.

16.9	Us	sage.	Words	used	hereir	$n, r \in$	egard	lless	of	the	numbe	er ar	nd geno	der
specifica	1115	y used	, will	be	deemed	and	cons	true	d to	inc	lude	any	other	number,
singular	or	plura	l, and	l any	other	geno	der a	s the	e co	ntex	t red	quire	es.	

Page 26

- 16.10 Governing Law. This Agreement will be governed by the laws of the State of Delaware in all respects.
- 16.11 No Third Party Benefited. It is not intended and this Agreement will not be construed to confer upon any person or entity not a party hereto any right or benefit. Nothing contained herein is intended or is to be construed as imposing any liability upon any party hereto, not expressed herein or required under the Act. Without limiting the foregoing, nothing contained in this Agreement will impose liability of the Company for debts or obligations of any Member or Manager or liability of any Member or Manager for debts and obligations of the Company.
- 16.12 Counterparts. This Agreement may be executed in one or more counterparts, each of which will be deemed to be an original and when any, but not necessarily the same, counterpart has been executed by all of the parties hereto, will constitute the binding agreement of such parties.
- 16.13 Entire Agreement. This Agreement, together with the Certificate of Formation of the Company, sets forth the entire agreement among the parties with regard to the subject matter hereof and supersedes all prior agreements with respect thereto whether written or oral. In the case of any conflict between this Agreement and the Certificate of Formation, the terms of the latter will control.
- 16.14 Binding Effect. Except as otherwise specified hereinabove, this Agreement will inure to the benefit of and be binding upon the heirs, successors, assigns and legal representatives of the parties hereto.

EXECUTED as of the date first appearing above.

THE COMPANY:

JETGLOBAL, LLC, a Delaware limited liability company

Ву:	/s/	Brian	Ν.	Hollnagel		
-					 Ма	nager

By:	/s/	John	Sawyer	
				 Manager

Page 27

MEMBERS:

BCI AIRCRAFT LEASING, INC.

By: /s/ Brian N. Hollnagel

Printed Name: Brian N. Hollnagel

\_\_\_\_\_

Title: President

\_\_\_\_\_

GLOBAL AIRCRAFT SOLUTIONS, INC.

By: /s/ John Sawyer

Printed Name: John B. Sawyer

\_\_\_\_\_

Title: President

-----

Reconditioning Cost Schedule

(REDACTED FOR CONFIDENTIALITY)

Global Aircraft Solutions and BCI Aircraft Leasing Form Joint Venture Corporation, Jetglobal, and Acquire 26 Boeing 737-200 Aircraft

TUCSON, Ariz.- September 1, 2005 - Global Aircraft Solutions, Inc. (OTCBB: GACF.OB) today announced that together with BCI Aircraft Leasing it has formed the joint venture corporation Jetglobal, LLC. This special purpose corporation has been formed by GACF and BCI to acquire and remarket commercial jet aircraft. Under the terms of the joint venture agreement, BCI will act as marketing lead, while GACF will act as the technical lead. BCI will own 70% of Jetglobal, while Global Aircraft will own 30%.

Following its incorporation, Jetglobal has acquired a fleet of 26 Boeing 737-200 Aircraft through Jetran International. All 26 aircraft were manufactured between 1983 and 1987 and are coming out of the Delta Airlines fleet. Of the 26 aircraft, 14 are parked and available for delivery, while 12 are still in service with Delta. The 12 aircraft still operated by Delta are scheduled to come off lease between September 2006 and October 2007. All rent revenue generated by the 12 aircraft still in service with Delta will devolve to Jetglobal until such time as the current leases are concluded or extended by Delta.

John Sawyer, President and Chief Operating Officer of GACF and Hamilton Aerospace, stated, "In my opinion, BCI has one of the very best management teams in the aircraft leasing industry. We are extremely pleased to be able to partner with this tremendously successful and innovative company. While we expect this first ex-Delta fleet transaction to have a significant positive impact on Global's 2005 and 2006 profitability, I am even more excited about the overall long-term potential of this strategic alliance with BCI."

Brian Hollnagel, President of BCI Aircraft Leasing commented, "Global Aircraft Solutions has built an excellent reputation for efficiently resolving all those complex technical issues that aircraft leasing companies routinely encounter when an aircraft comes off lease from one operator and then must go into service under totally different specifications for another airline operating in a different country. Our new strategic alliance with Global gives BCI the ability to incorporate high quality, cost-effective technical support into our proposals, which will be a valuable tool for helping us meet our customer's requirements."

Global Aircraft and Hamilton Aerospace Chairman & CEO Ian Herman added, "With this BCI joint venture, Global's aircraft trading business segment has now come of age, and no longer can be viewed as one-off, stand-alone transactions, but rather a viable business segment with reasonably predictable results that we intend to incorporate into future Company financial guidance and projections. The timing of the joint venture coincides with current sales activity, thus

minimizing the financial risk and cash requirements of this transaction. We expect Jetglobal to bring tremendous value to our shareholders."

# About BCI Aircraft Leasing

BCI Aircraft Leasing is fast becoming a recognized leader as on of the primary global aircraft leasing companies. The Company is able to provide airlines with unique and creative solutions to build, maintain and extract the greatest value from their fleet.

Headquartered in Chicago with a satellite office in Los Angeles and the Netherlands, BCI Aircraft Leasing is focused on the global market. The Company has aircraft in service with clients in Europe, North and South America and Africa.

However, the Company also has an active division targeting wide body aircraft for acquisition. This division's focus is to provide capital-intensive financial solutions to global operators of wide body aircraft.

Commonly, BCI specializes in such aircraft as the Boeing 737, 757, and A320 on lease to major domestic and international airlines. BCI's broad knowledge, strategic alliances, and extensive experience in this industry have resulted in BCI carving out a profitable niche that has allowed the Company to compete very effectively in the marketplace. BCI contacts provide extensive deal flow, allowing the Company to be very selective in its acquisition process, thus enabling the structure of transactions to accommodate internal operating hurdles of our airline clients as well as our investment objectives.

#### About Global Aircraft Solutions

Global Aircraft Solutions provides parts support and maintenance, repair and overhaul (MRO) services for large passenger jet aircraft to scheduled and charter airlines and aviation leasing companies. Hamilton Aerospace and World Jet, both divisions of Global Aircraft Solutions, operate from adjacent facilities comprising about 35 acres located at Tucson International Airport. These facilities include hangars, workshops, warehouses, offices and other buildings. Notable customers include debis AirFinance, BCI Aircraft Leasing, Q Aviation, Falcon Air Express, Jetran International, Goodrich Corporation, AAR, National Jet Systems, Pemco, San Antonio Aerospace, Pegasus Aviation, Shaheen Airlines, Iraqi Airways, and Aero California.

Global's website is located at www.globalaircraftsolutions.com. The Hamilton

Aerospace website is located at www.hamaerotech.com.

\_\_\_\_\_\_

Except for the historical information presented, the above statements are "forward-looking statements" within the meaning of the Private Securities

Litigation Reform Act of 1995 or regulations thereunder. These forward-looking statements are subject to risks and uncertainties, and actual results may differ materially. These risks include the economic health of the airline industry, demand for Global Aircraft Solutions' services, and competitive pricing pressures. In addition, other risks are detailed in Global's Form 10-KSB filed on April 4, 2005. These statements speak only as of above date, and Global disclaims any intent or obligation to update them.

#### Contact:

Global Aircraft Solutions Gordon Hamilton, (520) 294-3481 dhamilton@hamaerotech.com

Or

Alliance Advisors, LLC Alan Sheinwald, 914-244-0062 asheinwald@allianceadvisors.net

## EXECUTION VERSION

## AIRCRAFT SALE & PURCHASE AGREEMENT

Dated as of 2 September 2005

JETRAN, LLC

as Seller

and

## JETGLOBAL, LLC

#### as Purchaser

relating to up to twenty six (26) Boeing 737-200 Aircraft manufacturer's serial number - As specified herein engines manufacturer's serial numbers - As specified herein

## CONTENTS

Clav	ıse	Page
I.	Interpretation	1
2.	Representation~ And Warranties	1
3.	Agreement To Sell Arid Purchase	1
4.	Conditions Precedent	5

5.	Purchase Price		6					
6.	Pre-Delivery Inspection							
7.	Delivery							
8.	Aircraft Condit	ion & Warranties	8					
9.	Registration Fe	es	10					
10.	Indemnification		10					
11.	Further Provisi	ons	12					
12.	Governing Law		14					
13.	Brokers And oth	er Third Parties	14					
14.	Confidentiality		14					
SCHEDU	ULE 1		1					
	Part A	Description of Aircraft	1					
SCHED	ULE 2	Aircraft Documents	1					
SCHED	ULE 3	Definitions	1					
SCHEDU	ULE 4	Conditions Precedent	1					
	Part A	Seller Conditions Precedent	1					
	Part B	Purchaser Conditions Precedent	2					
ScHEDU	ULE 5	Representations and Warranties	1					
	Part A	Seller's Representations and Warranties	1					
	Part B	Purchaser's Representations and Warranties	1					
SCHED	ULE 6	Aircraft Bill of Sale	1					
SCHEDU	ULE 7	Form of Acceptance Certificate	1					

THIS AGREEMENT is made as of 2 September 2005

## AMONG:

JETRAN, LLC , a company organized in the State of Texas, United States of America, with its address at 12400 Hwy 281, North, Suite 150, San Antonio, Texas 78216 ("Seller"); and

JETGLOBAL, LLC, a company organized in the State of Delaware, United States of America, with its address at c/o BCI Aircraft Leasing, Inc., 330 North Wabash Avenue, Suite 2802, Chicago, Illinois 60611 ("Purchaser").

IT IS AGREED as follows.

#### I. INTERPRETATION

- 1.1 Definitions: Capitalized words and expressions are defined in Schedule 3.
- 1.2 Construction: Headings are to be ignored in construing this Agreement and unless the contrary intention is stated, a reference to:
  - (a) each of "Seller", "Purchaser" or any other Person includes any permitted successors and assignees;
  - (b) words importing the plural shall include the singular and vice versa;
  - (c) any document shall include that document as amended, novated, assigned or supplemented;
  - (d) a Clause or a Schedule is to a clause of or a schedule to this Agreement;
  - (e) any Law, or to any specified provision of any Law, is a reference to such Law or provision as amended, substituted or re-enacted.

#### 2. REPRESENTATIONS AND WARRANTIES

Seller represents and warrants to Purchaser in accordance with Part A of Schedule 5 and each of Purchaser represents and warrants to Seller in accordance with Part B of Schedule 5.

- 3. AGREEMENT TO SELL AND PURCHASE
- 3.1 Aircraft Status. On and subject to the terms of this Agreement:
  - (a) except as specified herein, the Aircraft will be sold by the Seller and purchased by the Purchaser on the Delivery Date in an "AS IS", "WHERE IS" condition; and
  - (b) Seller shall pass to Purchaser upon Delivery good and marketable title to the Aircraft clear of any and all liens other than Permitted Liens.

-1-

- 3.2 Delivery Documents: At Delivery of each Aircraft:
  - (a) If the Aircraft is owned by the Trust, Seller and Purchaser shall deliver to FM Counsel a duly executed Assignment and Assumption Agreement substantially in the form set forth in Schedule 9;
  - (b) If the Aircraft is not owned by the Trust,
    - (I) Seller shall deliver to Purchaser, the Bill of Sale for the Aircraft substantially in the form set forth in Schedule 6, duly executed by Seller;
    - (ii) Seller shall deliver to FM Counsel an FAA Bill of Sale for the

Aircraft duly executed by Seller;

- (iii) Purchaser shall deliver to FM Counsel the FAA Form 8050-1 Application for Registration duly executed by Purchaser for filing with the Air Authority;
- (b) Purchaser shall deliver to Seller the Acceptance Certificate substantially in the form set forth in Schedule 7, duly executed by Purchaser; and
- (c) Seller shall deliver such other documents as are reasonably necessary to transfer to Purchaser good and marketable title, free and clear of all liens in recordable form for the aircraft register of the jurisdiction of the Air Authority, and Purchaser shall deliver such other documents as are reasonably necessary to effect registration of the Aircraft in Purchaser's (or the Trustee's) name on the aircraft register of the jurisdiction of the Air Authority.

The documents referred to in this Clause 3.2 are defined for purposes of this agreement as the "Delivery Documents".

- 3.3 Security Interests: The Aircraft shall upon Delivery be free and clear of any Security Interests, liens other than Permitted Liens and claims of third parties with respect to Aircraft that are sold rather than transferred via a trust.
- 3.4 Passage of Title & Risk of Loss: Save as provided in the Bill of Sale or the Assignment and Assumption Agreement, title or Ownership Interest to the Aircraft shall pass to Purchaser when the Delivery Documents are delivered to Purchaser and the Seller, and filed with the FM as applicable. Risk of loss, destruction of, or damage to the Aircraft shall pass to Purchaser simultaneously with transfer of title.

-2-

- 3.5 Damage or Fault before Delivery: If before Delivery the Aircraft suffers material damage or a material fault occurs which (in either case) does not constitute an Event of Loss, then:
  - (a) Seller shall promptly notify Purchaser of the details of such damage or fault, (with proposals for repair where Seller reasonably considers repairs can be carried out before the Final Delivery Date);
  - (b) Purchaser shall in light of Seller's notice, notify Seller whether Purchaser is prepared to proceed subject to the damage or fault being repaired prior to the Final Delivery Date. If Purchaser is prepared to proceed on the basis of the repair, Seller shall procure the repair. However if (i) the repair is not so completed and the Aircraft is not delivered to Purchaser on or before the Final Delivery Date or (ii)

Purchaser reasonably declines to proceed on the basis of the repair of the Aircraft and notifies Seller, then no party hereto shall have any further obligation or liability to the other under this Agreement in respect of the Aircraft. In the event Purchaser is willing to proceed with the Purchase after repair of the Aircraft, then Seller shall proceed with such repairs and the parties hereto shall consummate this Agreement with respect to the repaired Aircraft.

- 3.6 Event of Loss Prior to Closing.
  - (a) Event of Loss. In the event that an Aircraft suffers an Event of Loss prior to the Closing with respect to such Aircraft, then, with effect from the date of such Event of Loss:
    - (i) this Agreement shall automatically terminate with respect to the Aircraft suffering such Event of Loss, and the Cash Purchase Price shall be recalculated to deduct the Allocated Purchase Price with respect to such Aircraft as set forth on Schedule 6 hereto;
    - (ii) Purchaser shall have no further obligation or liability to the applicable Seller with respect to such Aircraft (or the Ownership Interest relating to such Aircraft) under this Agreement,
    - (iii) Seller shall not have any further obligation or liability to
      Purchaser with respect to such Aircraft (or the Ownership
      Interest relating to such Aircraft) other than to refund to
      Purchaser the portion of the Deposit allocable to such Aircraft
      as set forth on Schedule 8 hereto; and
      - (iv) for the avoidance of doubt, each of the parties hereto expressly acknowledges and agrees that, notwithstanding the termination of this Agreement with respect to any particular Aircraft suffering an Event of Loss as provided in this Section 3.6(a), this Agreement shall remain in full force and effect, and each party hereto shall remain bound in all respects in accordance with the terms hereof, with respect to all other Aircraft (and the Ownership Interests relating thereto, as applicable) that do not suffer such Event of Loss.

-3-

(b) Incipient Event of Loss. If circumstances are such that with the passage of time andlor a relevant determination an Event of Loss might arise with respect to any Aircraft and those circumstances continue for more than twenty (20) days, then either Seller or Purchaser may at any time thereafter terminate this Agreement with respect to such Aircraft by giving notice to the other, in which case the provisions

of Section 3.6 (a) above shall apply from and after the delivery of such notice.

## 3.6A Additional Conditions to Closing.

The following are additional conditions to the obligation of the Purchaser to acquire each Aircraft or the related Ownership Interest:

- (a) Delta shall have delivered the Aircraft to Seller in accordance with the Omnibus Agreement.
- (b) Seller shall have notified Purchaser of delivery of the Aircraft to Seller and Purchaser shall have had at least fifteen (15) days to inspect the Aircraft and the Aircraft Documents to determine whether the Aircraft Documents have been delivered in compliance with the terms of the Omnibus Agreement and the Aircraft has been delivered in accordance with the return conditions in the Lease for such Aircraft. At the expiration of the 15 day inspection period, Purchaser shall provide written notice to Seller of its acceptance or rejection of the Aircraft.
- (c) With respect to the Category I Aircraft identified in Schedule 1, the Engines specified with respect to such Aircraft are on wing to such Aircraft.
- (d) With respect to the Category II Aircraft identified in Schedule 1, two of the Engines identified in the Engine Pool described in Schedule I are on wing to such Aircraft.

Provided, however, if any Engine required to be delivered on wing to any Aircraft cannot be delivered because of a total event of loss to such Engine Seller may offer a substitute engine for inspection and acceptance by Purchaser at its discretion, so long as such engine is of like model, value and utility.

## 3.7 Special Termination Provisions.

In the event Purchaser rejects an Aircraft that fails to meet the conditions precedent to Closing set forth in herein, then the following shall apply:

(i) this Agreement shall be deemed terminated with respect to the Aircraft so rejected by Purchaser, and the Cash Purchase Price shall be recalculated to deduct the Allocated Purchase Price with respect to such Aircraft as set forth on Schedule 8 hereto;

-4-

(ii) The deposit specified with respect to such Aircraft on Schedule 8

hereto will be promptly returned to Purchaser

- (iii) Purchaser shall have no further obligation or liability to the applicable Seller with respect to such Aircraft (or the Ownership Interest relating to such Aircraft) under this Agreement,
  - (iv) Seller shall nor have any further obligation or liability to Purchaser with respect to such Aircraft (or the Ownership Interest relating to such Aircraft) other than as speified in Section 3.6(a)(iii); and
  - (v) for the avoidance of doubt, each of the parties hereto expressly acknowledges and agrees that, notwithstanding the termination of this Agreement with respect to any particular Aircraft pursuant to this section, this Agreement shall remain in full force and effect, and each party hereto shall remain bound in all respects in accordance with the terms hereof, with respect to all other Aircraft (and the Ownership Interests relating thereto, as applicable) covered by this Agreement.

#### 4. CONDITIONS PRECEDENT

### 4.1 Seller Conditions:

Seller's obligation to sell the Aircraft to Purchaser shall be subject to fulfilment of the Seller Conditions Precedent on or before the Delivery Date (except to the extent that Seller agrees in writing in its absolute discretion to waive or defer any such condition).

The Seller Conditions Precedent have been inserted for Seller's benefit and may be waived in writing, in whole or in part and with or without conditions, by Seller without prejudicing its right to receive fulfilment of such conditions, in whole or in part, at any later time.

If any of the Seller Conditions Precedent remain outstanding on the Final Delivery Date and are not waived or deferred in writing by Seller, Seller may at any time after close of business in New York on the Final Delivery Date terminate its obligation to sell the Aircraft by notice to Purchaser, whereupon the rights and obligations of the parties hereunder shall cease and be discharged without further liability on the part of either Seller or Purchaser.

#### 4.2 Purchaser Conditions:

Purchaser's obligation to purchase the Aircraft shall be subject to fulfilment of each of the Purchaser Conditions Precedent on or before the Delivery Date (except to the extent that Purchaser agrees in writing in its absolute discretion to waive or defer any such condition).

The Purchaser Conditions Precedent have been inserted for Purchaser's benefit and may be waived in writing, in whole or in part and with or without conditions, by Purchaser without prejudicing its right to receive fulfilment of such conditions, in whole or in part at any later time.

If any of the Purchaser Conditions Precedent remain outstanding on the Final Delivery Date and are not waived or deferred in writing by Purchaser, Purchaser may at any time after close of business in New York on the Final Delivery Date terminate its obligation to purchase the Aircraft by notice, whereupon the rights and obligations of the parties hereunder shall cease and be discharged without further liability on the part of either Seller or Purchaser.

#### 5. PURCHASE PRICE

- 5.1 Amount: The purchase price for each Aircraft shall be the amount set forth with respect to such Aircraft in Schedule 8, or an aggregate of [CONFIDENTIAL PORTION DELETED AND FILED SEPARATELY WITH THE SECURITIES AND EXCHANGE COMMISSION] for all Aircraft (the "Purchase Price").
  - (a) Seller acknowledges receipt prior to execution of this agreement in the amount of [CONFIDENTIAL PORTION DELETED AND FILED SEPARATELY WITH THE SECURITIES AND EXCHANGE COMMISSION] (the "Deposit") which has been allocated to each Aircraft in accordance with Schedule 8.
  - (b) At the closing for each Aircraft at the time of Delivery, Purchaser will pay to the Seller the balance due on delivery of the Purchase Price for each Aircraft as set forth in Schedule 8, hereto.
- 5.3 Seller's Account: The Purchase Price shall be paid in Dollars to the "Jetran Escrow Account," account number [CONFIDENTIAL PORTION DELETED AND FILED SEPARATELY WITH THE SECURITIES AND EXCHANGE COMMISSION]
- 5.4 Payment of Taxes, Tax Indemnities:
  - (a) Without prejudice to Clause 7.3, the parties will co-operate so that the Delivery Location shall be in a jurisdiction where the imposition upon Seller or its subsidiary, affiliate, associated company, owner trust or assignee, General Electric Capital Corporation, GECAS, and/or Purchaser of any Taxes arising out of the sale of the Aircraft under this Agreement is minimized.
  - (b) Purchaser will indemnify and hold Seller, Seller's seller and its subsidiaries, affiliates, associated companies, owner trusts and assignees, General Electric Capital Corporation and GECAS ("Tax Indemnitees") harmless from any and all Taxes and expenses assessed against any Tax Indemnitee or the Aircraft or any part thereof by any Government Entity resulting from or arising in connection with the sale of the Aircraft hereunder, and any Taxes and expenses assessed against any Tax Indemnitee which are attributable to any payment made

- (i) any Taxes (including without limitation capital gains Taxes, minimum Taxes and doing business or franchise Taxes) imposed on the overall income, profits or gains of a Tax Indemnitee;
- (ii) any Taxes imposed as a result of Seller's failure to comply with this Agreement or non performance in relation to any applicable laws governing Seller's obligations hereunder; and
- (iii) any Taxes arising as a result of the gross negligence or willful misconduct of Seller.
- (C) All payments to be made by Purchaser under this Agreement shall be made in full without set off or counterclaim whatsoever and shall be made in full without any deduction or withholding whatsoever. If however a deduction or withholding for Taxes is required by law, each of Purchaser shall:
  - (i) ensure that the deduction or withholding does not exceed the minimum amount legally required;
  - (ii) immediately pay to Seller (or such other party, as the case may be) such additional amount so that the net amount received by Seller or such party will equal the full amount which would have been received by it had no such deduction or withholding been made;
  - (iii) pay to the relevant taxation authority or other authorities within the time allowed by law the full amount of the deduction or withholding (including, but without prejudice to the generality of the foregoing, the full amount of any deduction or withholding from any additional amount paid pursuant to this sub-clause); and
    - (iv) provide Seller or such other party within the period for payment permitted by the relevant law with an official receipt of the relevant taxation authorities for all amounts so deducted or withheld or if such receipts are not issued by the taxation authorities concerned, a certificate of deduction or equivalent evidence of the relevant deduction or withholding.

## 6. PRE-DELIVERY INSPECTION

6.1 Subject to satisfactory completion of Purchaser's pre-delivery inspection of the Aircraft in accordance with Section 3.6A(b), Purchaser shall execute

#### 7. DELIVERY

- 7.1 Delivery: Subject to satisfaction (or waiver or deferral with the agreement in writing of Seller) of the Seller Conditions Precedent, Seller shall on the Delivery Date: i) transfer title of the Aircraft through the execution and delivery of the applicable Delivery Documents, ii) deliver the Aircraft Documents to Purchaser; and iii) transfer care, custody and control of the Aircraft to Purchaser.
- 7.2 Delivery Data: The parties currently anticipate that Delivery will take place on the Expected Delivery Date and shall each use reasonable efforts so that Delivery does then take place but in any event Delivery shall occur no later than close of business New York time on the Final Delivery Date.
- 7.3 Delivery Location: The Bill of Sale for the Aircraft shall be delivered to Purchaser while the Aircraft is located in one of the following locations (the "Delivery Location"):
  - (a) the Expected Delivery Location; or
  - (b) with the agreement of the parties, another jurisdiction provided
    - (i) the Lex Situs Opinion is prepared and issued at Purchaser's expense to Seller and Purchaser on or prior to Delivery; and
    - (ii) the parties are satisfied that no Taxes will be imposed upon any Tax Indemnitee, Purchaser or the Aircraft as a result of the transfer of title to the Aircraft while the Aircraft is located in such jurisdiction, other than any Taxes which Seller or Purchaser may agree in writing to bear.

Seller shall keep Purchaser advised as to the intended flight schedule for the Aircraft during the period prior to the Delivery Date.

- 7.4 Tender and Acceptance of Delivery: On the Delivery Date, subject to this Agreement, Seller shall tender the Aircraft for Delivery and Purchaser shall accept delivery of the Aircraft by executing and delivering the Acceptance Certificate to Seller.
- 8. AIRCRAFT CONDITION & WARRANTIES
- 8.1 Acceptance Certificate: Delivery of the Acceptance Certificate by Purchaser to Seller shall be conclusive proof that Purchaser has examined and

investigated the Aircraft and that it is acceptable and in every way satisfactory to Purchaser.

- 8.2 Reserved.
- 8.3 Reserved.
- 8.4 Limitation on Warranties.

-8-

- EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THIS AGREEMENT OR IN THE (a) OTHER TRANSFER DOCUMENTS TO THE CONTRARY, AS BETWEEN THE APPLICABLE SELLER AND PURCHASER, PURCHASER SHALL UNCONDITIONALLY ACCEPT EACH OWNERSHIP INTEREST AND EACH SALE AIRCRAFT, IF ANY, "AS IS "'WHERE IS" AND 'WITH ALL FAULTS." THE REPRESENTATIONS, WARRANTIES AND COVENANTS SET FORTH IN THIS AGREEMENT AND T~E OTHER TRANSFER DOCUMENTS ARE EXCLUSIVE AND IN LIEU OF ALL OTHER REPRESENTATIONS, WARRANTIES AND COVENANTS OF SELLER OF ANY KIND WHATSOEVER, WHETHER WRITTEN, ORAL, EXPRESS OR IMPLIED, AND NO SELLER HAS MADE AND NO SELLER SHALL BE DEEMED TO HAVE MADE, AND EACH SELLER HEREBY DISCLAIMS, (I) ANY REPRESENTATION, WARRANTY OR COVENANT (EXCEPT THE REPRESENTATIONS, WARRANTIES AND COVENANTS SET FORTH IN THIS AGREEMENT AND THE OTHER TRANSFER DOCUMENTS TO WHICH SUCH SELLER IS A PARTY) AS TO THE TITLE, AIRWORTHINESS, CONDITION, DESIGN, VALUE, OPERATION, MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE OR USE OF ANY AIRCRAFT (OR ANY PART THEREOF), THE FREEDOM OF ANY AIRCRAFT (OR ANY PART THEREOF) FROM ANY LATENT OR OTHER DEFECT (WHETHER OR NOT DISCOVERABLE) OR ANY ACTUAL OR ALLEGED INFRINGEMENT OF ANY PATENT, TRADEMARK, COPYRIGHT OR THE LIKE, OR THE COMPLIANCE OF ANY AIRCRAFT (OR ANY PART THEREOF) WITH ANY APPLICABLE LAWS OR REGULATIONS , (II) ANY REPRESENTATION, WARRANTY OR COVENANT WITH RESPECT TO THE TAX OR ACCOUNTING TREATMENT OF ANY OWNERSHIP INTEREST OR ANY TRUST ESTATE, AND (III) ALL OTHER REPRESENTATIONS, WARRANTIES AND COVENANTS, WHETHER WRITTEN, ORAL, EXPRESS OR IMPLIED (EXCEPT THE REPRESENTATIONS AND WARRANTIES SET FORTH IN THIS AGREEMENT AND THE OTHER TRANSFER DOCUMENTS TO WHICH SUCH SELLER IS A PARTY); IT BEING UNDERSTOOD THAT ALL SUCH FOREGOING DISCLAIMED RISKS, AS BETWEEN SELLER AND PURCHASER, ARE TO BE BORNE SOLELY BY PURCHASER.
- (b) Notwithstanding anything to the contrary set forth elsewhere in this Agreement or in any of the other transfer documents, in the event of any breach by any Seller of the title warranties set forth in Schedule 5 Part A in respect of any Sale Aircraft, the maximum liability of such Seller in respect of any such breach shall be limited to the amount equal to the Allocated Purchase Price with respect to such Aircraft, as set forth in Schedule 8 hereto, and in no event shall any such Seller have any obligation or liability to Purchaser or Purchaser's Designee in respect of such breach other than to pay to

#### 9. REGISTRATION FEES

Purchaser shall bear any duties or fees payable to the Air Authority in connection with the transfer of title to the Aircraft from Seller to Purchaser.

#### 10. INDEMNIFICATION.

Indemnification of Purchaser. Seller agrees to indemnify, defend and hold harmless Purchaser and each Purchaser's Designee (if any) and their respective officers, employees or agents (collectively, "Purchaser Indemnitees") from and against any and all Losses suffered or incurred by any Purchaser Indemnitee to the extent the Losses are based upon or arise out of (I) the purchase, manufacture, ownership, possession, registration, performance, transportation, management, sale, control, inspection, use or operation, design. condition, testing, delivery, leasing, maintenance, repair, service, modification, overhaul, replacement, removal or redelivery of such Aircraft, or any loss of or damage to such Aircraft, or otherwise in connection with such Aircraft or relating to loss or destruction of or damage to any property, or death or injury to any person caused by, relating to or arising from or out of (in each case whether directly or indirectly) any of the foregoing matters, in each case to the extent occurring or attributable to the period of time prior to the Effective Time on the applicable Closing Date for such Aircraft; (ii) any design, article or material in the Aircraft or the operation or use thereof constituting an infringement of patent, copyright, trademark, design or other proprietary right or a breach of any obligation of confidentiality owed to any person to the extent that any Losses attributable to the matters described in this sub-clause (ii) arise out of any act, omission, event or circumstance occurring prior to the Effective Time on the applicable Closing Date; (iii) the Ownership Interest relating to such Aircraft in respect of or relating to the period of time prior to the Effective Time on the applicable Closing Date for such Aircraft; (iv) any breach by Seller of any representation or warranty of Seller contained herein or in any other Transfer Document; (v) any failure of Seller to pay, perform and comply with any covenant, duty or obligation of Seller contained in any Applicable Document relating to such Aircraft prior to the Effective Time on the applicable Closing Date; or (vi) any failure of Seller to pay, perform or comply with any covenant, duty or obligation of Seller contained herein or in any other Transfer Document; provided that the liability of the applicable Seller for any obligation under

clauses (i), (iii), (iii) and (v) shall be solely for Losses which arise directly from a third party claim against a Purchaser Indemnitee and no Seller shall have any obligations under clauses (i), (ii), (iii) or (v) above with respect to any Tax or (in the case of any other Loss) if and to the extent that Lessee is obligated under any applicable Lease or otherwise to provide such indemnification to the applicable Purchaser Indemnitee. The indemnity set forth in this Section 10(a) shall not apply to any Loss (1) which is attributable to the willful misconduct or gross negligence of Purchaser, any

-10-

Purchaser's Designee or any of their respective employees, servants or agents, (2) which results from the breach by Purchaser or any Purchaser's Designee of any representation, warranty or covenant made herein, in any of the other Transfer Documents or in any Applicable Document relating to any Aircraft, (3) which are ordinary or usual operating or overhead expenses of Seller (except to the extent the same arise on the occurrence of any breach by Purchaser or Purchaser's Designee of its obligations under this Agreement or the other applicable Transfer Documents), (4) which are required to be borne Seller in accordance with any other express provision contained in this Agreement or any other Transfer Document.

Indemnification of Sellers. The Purchaser agrees to indemnify, defend (b) and hold harmless Seller, and each Seller's Designee and their shareholders, subsidiaries, affiliates, partners, contractors, officers, directors, representatives, agents and employees (collectively, "Seller Indemnitees") from and against any and all Losses suffered or incurred by any Seller Indemnitee, to the extent the Losses are based upon or arise out of (i) the purchase, manufacture, ownership, possession, registration, performance, transportation, management, sale, control, inspection, use or operation, design, condition, testing, delivery, leasing, maintenance, repair, service, modification, overhaul, replacement, removal or redelivery of such Aircraft, or any loss of or damage to such Aircraft, or otherwise in connection with such Aircraft or relating to loss or destruction of or damage to any property, or death or injury to any person caused by, relating to or arising from or out of (in each case whether directly or indirectly) any of the foregoing matters, in each case to the extent occurring or attributable to the period of time after the Effective Time on the applicable Closing Date for such Aircraft, (ii) any design, article or material in the Aircraft or the operation or use thereof constituting an infringement of patent, copyright, trademark, design or other proprietary right or a breach of any obligation of confidentiality owed to any person to the extent that any Losses attributable to the matters described in this sub-clause (iii) arise out of any act, omission, event or

circumstance occurring after the Effective Time on the applicable Closing Date; (iii) the Ownership Interest relating to such Aircraft in respect of or relating to the period of time from and after the Effective Time on the applicable Closing Date for such Aircraft; (iv) any breach by the Purchaser or any Purchaser's Designee of any representation or warranty of the Purchaser or such Purchaser's Designee contained herein or in any other Transfer Document; or (iv) any failure of the Purchaser to pay, perform and comply with any covenant, duty or obligation of the Purchaser contained in any Applicable Document relating to such Aircraft to the extent expressly assumed by the Purchaser pursuant to the applicable Assignment and Assumption Agreement; (v) any failure of the Purchaser or any Purchaser's Designee to pay, perform or comply with any covenant, duty or obligation of the Purchaser or such Purchaser's Designee contained herein or in any other Transfer Document;; provided that the liability of the Purchaser for any obligation under clauses (i), (ii), (iii) and (iv) shall be solely for Losses which arise directly from a third party claim against a Seller Indemnitee and the Purchaser shall have no obligations under clauses (i), (ii), (iii) or (iv) above with respect to any Tax or (in the case of any other Loss) if and to the extent that Lessee is obligated under any applicable Lease or otherwise to provide such indemnification to the applicable Seller Indemnitee. The indemnity set forth in this Section 10(b) shall not apply to any Loss (1) which is attributable to the willful misconduct or gross negligence of any Seller or any of their respective employees, servants or agents, (2) which results from the breach by

-11-

Seller of any representation, warranty or covenant made herein, in any of the other Transfer Documents relating to any Aircraft, (3) which are ordinary or usual operating or overhead expenses of Seller (except to the extent the same arise on the occurrence of any breach by Purchaser or Purchaser's Designee of its obligations under this Agreement or the other applicable Transfer Documents), or (4) which are required to be borne by Seller in accordance with any other express provision contained in this Agreement or any other Transfer Document.

#### ii. FURTHER PROVISIONS

- 11.1 Benefit of Agreement: Neither party shall assign or transfer all or any of its rights and/or obligations under this Agreement without the prior written consent of the other party.
- 11.2 Counterparts: This Agreement may be executed in any number of separate counterparts and each counterpart shall when executed and delivered be an original document but all counterparts shall together constitute one and the same instrument.

- 11.3 Waivers and Variation: Rights of a party, arising under this Agreement or the general law, shall not be waived or varied unless done so expressly in writing and only then in that specific case, on that specific occasion and on any terms specified.
- 11.4 Third Party Rights: A person who is not a party to this Agreement has no direct right to enforce any term of this Agreement nor to object or be consulted about any amendments to this Agreement.
- 11.5 Notices: Any notice in connection with this Agreement shall be given in writing and in English. A notice shall be delivered personally or by post, email or facsimile as detailed below (or as otherwise notified by the receiving party from time to time). A notice shall be deemed received if posted, three (3) days after it is mailed; if sent by hand or courier, when it is delivered; if faxed, when the fax is sent with a dear transmission report; if by email, when received into the 'In Box' of the recipient and a "read receipt" generated:

to Seller at:

JETRAN, LLC 12400 Hwy 281, North, Suite 150 San Antonio, Texas 78216

Attention: Douglas Jaffe Fax: (210) 495-7799

Email: jaffe@jetraninterntional.com

-12-

to Purchaser at:

JetGIobal, LLC C/O BCI Aircraft Leasing, Inc. One IBM Plaza 330 North Wabash Avenue, Suite 2802 Chicago, Illinois 60611

Attention: Brian N. Hollnagel Fax: (312) 329-1250

Email: brian@bciaircraft.com

with copies to:

Global Aircraft Solutions, Inc. 6901 South Park Avenue Tucson, Arizona 85706 Attention:

Fax: (520) 741-1430

Email:

Brian D. Fitzgerald, Esq. Rothgerber Johnson & Lyons, LLP 1200 17th Street, Suite 3000 Denver, Colorado 80202

Fax: (303) 623-9222

Email: bfitzgerald@rothgerber.com

11.6 Invalidity of any Provision: If any part of this Agreement becomes invalid, illegal or unenforceable under any applicable law, the validity, legality and enforceability of the remaining provisions shall not in any way be affected.

- 11.7 Entire Agreement: This Agreement together with the schedules attached hereto constitutes the entire agreement between the parties hereto in relation to the sale and purchase of the Aircraft and supersedes all previous proposals, representations, agreements and other written and oral communications in relation thereto.
- 11.8 Costs and Expenses: Except where this Agreement states differently, each party shall bear its own fees, costs and expenses arising out of or connected with this Agreement. For the avoidance of doubt, Seller shall not be responsible for any broker's fees, commissions or expenses relating to the sale of the Aircraft.

#### 12. GOVERNING LAW

This Agreement in all respects shall be governed by, and construed in accordance with, the laws of the State of New York. The U.N. Convention on Contracts for the International Sales of Goods is not applicable to this Agreement and all of its terms must be construed in accordance with the law applicable to domestic transactions in the State of New York.

-13-

#### 13. BROKERS AND OTHER THIRD PARTIES

- 13.1 No Brokers: Each party represents and warrants to the other that it has not paid, agreed to pay or caused to be paid directly or indirectly in any form, any commission, percentage, contingent fee, brokerage or other similar payments of any kind, in connection with the establishment or operation of this Agreement, to any Person (other than fees payable by each party to its legal advisers or any management or related fees and expenses which may be paid or payable to Seller).
- 13.2 Indemnity: Each party agrees to indemnify and hold the other harmless from and against any and all claims, suits, damages, costs and expenses

(including, but not limited to reasonable attorneys' fees) asserted by any agent, broker or other third party for any commission or compensation of any nature whatsoever based upon this Agreement or the Transaction Documents or the Aircraft, if such claim, suit, damage, cost or expense arises out of any breach by the indemnifying party, its employees or agents of Clause 13.1.

#### 14. CONFIDENTIALITY

The parties will hold the financial terms of this Agreement, any non-public financial information concerning any of the parties and all non-public information obtained pursuant to the requirements hereof which has been identified as confidential by any party, except as required or requested by any law, rule or regulation of any governmental agency or representative thereof, delivered in response to legal process or delivered to independent auditors and counsel to the disclosing party. It is also understood that any party may disclose to any and all persons, without limitation of any kind, the U.S, federal income tax treatment and U.S. federal income tax structure of the transactions described herein and all materials of any kind (including opinions or other tax analyses) relating to such tax treatment and tax structure, provided that if any document or similar item contains information concerning U.S. federal income tax treatment or U.S. federal income tax structure, as well as other information, this understanding will apply only to such portions of the document or other similar items that relate to such tax treatment or such tax structure.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK.]

- 14 -

## EXECUTION PAGE

IN WITNESS whereof this Agreement has been signed on the day and year first above written

JETRAN, LLC

/s/ M. D. Jaffe, Jr. By: \_\_\_\_\_

Name: M. D. Jaffe, Jr. Name:

Title: C.E.O.

\_\_\_\_\_

Purchaser

JETGLOBAL, LLG

By: /s/ John B. Sawyer

\_\_\_\_\_

Name: John B. Sawyer

Vice President/Manager \_\_\_\_\_

-S-

### SCHEDULE 1

## Part A Description Of Aircraft CATEGORY 1 AIRCRAFT \_\_\_\_\_

	LESSEE	U.S.	NO. I	NO. 2	SCHEDULED
	SHIP		ESN	ESN	EXPIRY
MSN NO.		REG.			DATE
		NO.			
23079	307	N307DL	709163	709143	30-Sep-05
23084	312	N3I2DL	709118	709130	TBA

23085	313	N3I3DL	709123	709160	TBA
23087	315	N3I5DL	709120	709165	TBA
23088	316	N3I6DL	709125	709195	TBA
23089	317	N3I7DL	709132	709164	15-Sep-05
23090	318	N3I8DL	709166	709183	30-Sep-05
23092	320	N32ODL	709115	709159	TBA
23093	321	N321DL	709184	709152	30-Sep-05
23094	322	N322DL	709133	709129	30-Sep-05
23099	327	N327DL	717159	709186	30-Sep-05
23100	328	N328DL	709180	709178	TBA
23608	381	N38IDL	709181	717160	02-Sep-05
23609	382	N382DL	709150	709127	02-Sep-05

## CATEGORY 2 AIRCRAFT

	LESSEE SHIP	U.S.	NO.1 ESM	NO.2 ESN	SCHEDULED EXPIRY
MSN NO.		REG.			DATE
		NO.			
23073	301	N3OIDL	*	*	30-Sep-06
23074	302	N3020L	*	*	31-Oct-06
23075	303	N3O3DL	*	*	30-Nov-06
23076	304	N304DL	*	*	31-Dec-06
23077	305	N305DL	*	*	31-Jan-07
23078	306	N3060L	*	е	28-Fe1;-07
23080	308	N308DL	*	*	31-Mar-07
23081	309	N3O9DL	*	*	31-Mar-07
23086	314	N3I4DA	*	*	31-May-07
23098	326	N326DL	*	*	31-Oct-07
23101	329	N329DL	*	*	31-Dec-06
23102	330	N32ODL	*	*	31-Oct-07

<sup>\*</sup>Except as provided in this Agreement, each Category 2 Aircraft shall be delivered with two Engines from the following pool:

700087	709124	709144	709154	709158	709185
709114	709128	709149	709155	709167	709194
709119	709131	709151	709156	709169	717154
709122	709135	709153	709157	709179	717163

All engines are [specify manufacturer and type], and have rated takeoff horsepower of 750 or more.

#### SCHEDULE 2

#### AIRCRAFT DOCUMENTS

With respect to each Aircraft other than any Aircraft delivered on lease (if any), the Seller shall have exercised good faith efforts to cause Lessee to deliver to Purchaser all of the Aircraft Documents in respect of such Aircraft and the Aircraft Documents so delivered to Purchaser will be in a format substantially similar to the Aircraft Documents relating to any one of the Aircraft bearing manufacturer's serial numbers 23084, 23085, 23100 and 23608, which were provided to Purchaser for inspection prior to the execution and delivery of this Agreement. Seller and Purchaser hereby expressly acknowledge and agree that in the event that the Aircraft Documents tendered to Purchaser by Lessee in respect of any Aircraft other than the ones identified by manufacturer's serial number in the preceding sentence or any Aircraft delivered on lease, do not meet the standard specified in the immediately preceding sentence, Purchaser shall have the option of either (I) accepting the Aircraft with such Aircraft Documents as shall have been tendered to Purchaser by Lessee, in which case the parties shall proceed to close the sale and purchase of such Aircraft in accordance with the this agreement, rejecting such Aircraft for failure to satisfy the condition precedent set forth in this Schedule 4, in which case the parties respective rights and obligations in respect of such Aircraft and this Agreement shall be governed by Section 3.7 above.

## SCHEDULE 3 DEFINITIONS

- "Acceptance Certificate" means a certificate of acceptance substantially in the form of Schedule 7;
- "Air Authority" means the civil aviation authority of the State of Registration;
- "Aircraft" means the aircraft described in Schedule I (which term includes, where the context admits, a separate reference to all Engines, Parts and Aircraft Documents);

"Aircraft Documents" means the documents specified in Schedule 2, and all additions, renewals and replacements made thereto prior to Delivery, to the extent that Seller has acquired title thereto;

"Bill of Sale" means each of the FAA Form 8050-2 bill of sale and the warranty bill of sale substantially in the form of Schedule 6, executed by Seller in respect of the Aircraft (collectively, the "Bills of Sale");

"Business Day" means a day (other than a Saturday or Sunday) on which banks are open for business in New York;

"Delivery" means the transfer of title of the Aircraft and transfer of care, custody and control of the Aircraft by Seller to Purchaser hereunder;

"Delivery Condition Requirements" means the requirements set out in Part B of Schedule 1;

"Delivery Date" means the date upon which closing of the acquisition of the Aircraft or Ownership Interest by Purchaser occurs.

- "Delivery Documents" is defined in Clause 3.2;
- "Delivery Location" is defined in Clause 7.3(a);
- "Delta'~ means Delta Air Lines, Inc.
- "Engines" means the engines specified in Schedule 1, together with all equipment and accessories belonging to, installed in, or appurtenant to, such engines;

"Event of Loss" means with respect to the Aircraft (including for the purposes of this definition the Airframe):

- (a) the actual or constructive total loss of the Aircraft (including any damage to the Aircraft which results in an insurance settlement on the basis of a total loss, or requisition for use or hire which results in an insurance settlement on the basis of a total loss); or
- (b) the Aircraft being destroyed, damaged beyond economic repair or permanently rendered unfit for normal use for any reason whatsoever; or
- (c) the requisition of title or other compulsory acquisition of title for any reason of the Aircraft by the government of the State of Registration or any other authority (whether de jure or de facto>; or
- (d) the hijacking, theft, disappearance, condemnation, confiscation, seizure, detention or requisition for use or hire of the Aircraft which deprives Seller or any Person permitted by Seller to have possession and/or use of the Aircraft of its possession and/or use for (i) more than 15 days (or 30 days, in the case of requisition for use

or hire by the government of the State of Registration) or (ii) if earlier, a period ending on the Final Delivery Date;

"Expected Delivery Date" has the meaning ascribed in the Omnibus Agreement

"Expected Delivery Location" is the location for delivery of the Aircraft described in the Omnibus Agreement provided that Purchaser will not be obligated to accept delivery of the Aircraft over international waters.

"FAA" means the United States Federal Aviation Administration.

"FAA Counsel" means Daugherty Fowler Peregrin Haught & Jenson, the address of which is 204 N. Robinson, Suite 900, Oklahoma City, Oklahoma 73102, facsimile no. (405) 232-0865.

"GECAS" means either or both GE Capital Aviation Services, Inc. and GE Commercial Aviation Services Limited, and their successors and assigns;

## "Government Entity" means:

- (a) any national government, political subdivision thereof, or local jurisdiction therein;
- (b) any instrumentality, board, commission, court, or agency of any of the above, however constituted; and
- (c) any association, organization, or institution of which any of the above is a member or to whose jurisdiction any thereof is subject or in whose activities any of the above is a participant;

"Law" includes (a) any statute, decree, constitution, regulation, order, judgement or other directive of any Government Entity; (b) any treaty, pact, compact or other agreement to which any Government Entity is a signatory or party; (c) any judicial or administrative interpretation or application of any Law described in (a) or (b) above; and (d) any amendment or revision of any Law described in (a), (b) or (c) above;

"Lease" means a lease between Delta and a seller under the Omnibus Agreement with respect to an Aircraft.

"Lex Situs Opinion" means an opinion of counsel acceptable to Seller and Purchaser in the jurisdiction in which the Aircraft is located at Delivery, in form and substance satisfactory to both parties;

"Losses" means losses, liabilities, claims, proceedings, penalties, judgments, damages, costs and expenses;

"Manufacturer" means Boeing;

- "Omnibus Agreement" means the Omnibus Sale Agreement dated August 31, 2005 between GECAS and Seller.
- "Ownership Interest" means the owner participant's interest in the Trust with respect to any Aircraft.
- "Part" means, whether or not installed on the Aircraft, any component, furnishing or equipment (other than a complete Engine) furnished with the Aircraft on the Delivery Date;
- "Permitted Lien" means any Security Interest created by or resulting from debts or liabilities or actions of Purchaser;
- "Person" means any individual person, corporation, partnership, firm, joint stock company, joint venture, trust, estate, unincorporated organization, association, Government Entity, or organization or association of which any of the above is a member or a participant;
- "Purchase Price" is defined in Clause 5.1;
- "Purchaser Conditions Precedent" means the conditions set out in Part B of Schedule 4;
- "Security Interest" means any mortgage, charge, pledge, lien, encumbrance, assignment, hypothecation, right of set-off or any other agreement or arrangement having the effect of conferring security;
- "Seller Conditions Precedent" means the conditions specified in Part A of Schedule 4; "State of Registration" means United States of America; "Taxes" means any and all present and future taxes, duties, withholdings, levies, assessments, imposts, fees and other governmental charges of all kinds (including without limitation any value added or similar tax and any stamp, documentary, registration or similar tax), together with any penalties, fines, surcharges and interest thereon and any additions thereto;
- "Transaction Documents" means this Agreement, the Bills of Sale, the Acceptance Certificate and any agreement amending or supplementing any of the foregoing documents;
- "Trust" means any trust holding title to an Aircraft and the lessor's interest in the Lease of that Aircraft.
- "Trustee" means the trustee under any trust.
- "US\$" and "Dollars" means the lawful currency of the United States of America, and (in relation to all payments in dollars to be made under this Agreement) same day funds.

## SCHEDULE 4 CONDITIONS PRECEDENT

## Part A Seller Conditions Precedent

- I. Seller shall have received each of the following documents and evidence:
  - o an opinion of counsel regarding due execution of this Agreement and other relevant Transaction Documents by Purchaser and other matters, in form and substance satisfactory to Seller;
    - the Acceptance Certificate duly executed by Purchaser;
    - if applicable, the Assignment and Assumption Agreement;
  - o if Delivery occurs while the Aircraft is not located in Expected Delivery Location, the Lex Situs Opinion duly signed by the counsel providing the same;
- 2. Seller shall have received on or before the Delivery Date, the Purchase Price;
- 3. Seller shall be satisfied that the Delivery Location, and the arrangements described in Clause 7, do not give rise to any Taxes, other than Taxes which Purchaser or Seller shall have agreed in writing to bear;
- 4. Purchaser shall have pre-positioned in escrow with FAA Counsel the documents required to be delivered by Purchaser to FAA counsel in accordance with the Agreement;
- 5. The representations given by each of the Purchaser in Part B of Schedule 5 being true and accurate on the Delivery Date;
- 6. Purchaser shall not be in material default of its obligations under this Agreement; and
- 7. No change having occurred after the date of this Agreement in any

applicable Law which would make it illegal for Seller to perform any of its obligations under this Agreement (and any other documents to be entered into pursuant hereto); provided that if any such change has occurred the parties may but shall not be obliged to use all reasonable co-operative endeavors to restructure the transaction contemplated by such documents so as to avoid the aforementioned illegality.

## Part B Purchaser Conditions Precedent

- 1. Purchaser shall have received each of the following documents and evidence:
  - . if applicable, the Bills of Sale in respect of the Aircraft;
    - if applicable, the Assignment and Assumption Agreement;
  - o an opinion of counsel regarding due execution of this Agreement and other relevant Transaction Documents by Seller and other matters, in form and substance satisfactory to Purchaser;
  - o if Delivery occurs while the Aircraft is not located in Expected Delivery Location, the Lex Situs Opinion duly signed by the counsel providing the same;
  - o if the Aircraft is currently under lease, an insurance certificate and broker's opinion in form and substance satisfactory to Purchaser, naming the Purchaser and party providing funding to the Purchaser as loss payee and additional insured, together with a consent to assignment of the owner participant's interest in the trust or sale of the Aircraft to Purchaser, as appropriate.
- 2. Reserved.
- 3. Purchaser shall be satisfied that the Delivery Location, and the arrangements described in Clause 7, do not give rise to any Taxes, other than any taxes which Seller or Purchaser shall have agreed in writing to bear;
- 4. Purchaser shall have pre-positioned in escrow with FM Counsel the documents required to be delivered by Purchaser to FAA counsel in accordance with the Agreement;
- 5. The Aircraft shall not have suffered an Event of Loss on the Delivery Date;
- 6. The representations given by Seller in Part A of Schedule 5 being true and accurate on the Delivery Date;

- 7. No change having occurred after the date of this Agreement in any applicable Law which would make it illegal for Purchaser to perform any of its obligations under this Agreement (and any other documents to be entered into pursuant hereto); provided that if any such change has occurred the parties may but shall not be obliged to use all reasonable co-operative endeavors to restructure the transaction contemplated by such documents so as to avoid the aforementioned illegality.
- 8. The Seller not being in material default of its obligations under this Agreement.

## SCHEDULE 5 REPRESENTATIONS AND WARRANTIES

# Part A Sellers Representations And Warranties

- 1. General Representations and Warranties: Seller represents and warrants to Purchaser as follows
  - o Seller duly exists and has the power to enter into and implement the transactions contemplated by the Transaction Documents to which it is a party;
  - o the execution, delivery and performance of the Transaction Documents to which it is a party have been duly authorized by all necessary action on the part of Seller;
  - o the Transaction Documents to which it is a party Constitute legal, valid and binding obligations of Seller;
  - o each consent required by Seller to authorize, or required by it in connection with the execution, delivery, performance, legality, validity or enforceability of the Transaction Documents to which it is a party has been obtained and is in full force and effect, and there is no default in the observance or performance of any of the conditions and restrictions (if any) imposed on or in connection therewith; and
  - o the execution, delivery and performance by Seller of the Transaction Documents to which it is a party will not (i) conflict with, or result in any material breach of, any of the terms of 1 or constitute a default under, any agreement or document to which it is a party or by which it or any of its property or assets may be bound or (ii) contravene or conflict with the provisions of its constitutive documents.

## Part B Purchaser's Representations And Warranties

With respect to itself, Purchaser represents and warrants to Seller that the following statements are tme and accurate:

- O Purchaser duly exists under the laws of its state of organization in the United States of America and has the power to enter into and implement the transactions contemplated by the Transaction Documents to which it is a party;
- o the execution, delivery and performance of the Transaction Documents to which it is a party have been duly' authorized by all necessary corporate action on the part of Purchaser;
- o the Transaction Documents to which it is a party constitute legal, valid and binding obligations of Purchaser;
- o each consent required by Purchaser to authorize, or required by it in connection with the execution, delivery, performance, legality, validity or enforceability of the Transaction Documents to which it is a party has been obtained and is in full force and effect, and there is no default in the observance or performance of any of the conditions and restrictions (if any) imposed on or in connection therewith; and
- o the execution, delivery and performance by Purchaser of the Transaction Documents to which it is a party will not (I) conflict with, or result in any material breach of, any of the terms of, or constitute a default under any agreement or document to which it is a party or by which it or any of its property or assets may be bound or (ii) contravene or conflict with the provisions of its constitutive documents.

## SCHEDULE 6 AIRCRAFT BILL OF SALE

One (1) B737-200 Aircraft manufacturer's serial number

WHEREAS, JETRAN, LLC, as seller (the "Seller"), is the owner of the aircraft, engines, equipment and documents described below (hereinafter referred to as the "Aircraft1'):

0	B737-200A	aircraft bearing	g manuf	acturer	's serial	number	with
	two Pratt	& Whitney model	JT8D-	engines	bearing	manufacturer's	serial
	numbers	and					

o all equipment, accessories and parts belonging to, installed in or appurtenant to such aircraft or engines; and

as further described and defined in that certain Aircraft Sale & Purchase Agreement dated as of August \_\_ 2005 among Jetran, LLC, as Seller and JetGlobal, LLC, as Purchaser (the "Aircraft Sale and Purchase Agreement"). Capitalized words used herein shall have the meaning given to such words in the Aircraft Sale and Purchase Agreement.

Now, THE	REFORE,	for good	and valuable	le considera	ation the	receipt a	nd suffic	iency
of which	is here	eby ackno	wledges, Se	ler does he	ereby sell	L, grant,	convey,	
transfer	and de	liver and	set over un	nto		a		
			("Purch	naser'), ali	l of Selle	er's right	, title a	nd
interest	in and	to the A	ircraft, sub	ject to Pe	rmitted Li	lens.		

Seller hereby warrants to Purchaser that there is hereby conveyed to Purchaser good and marketable title to the Aircraft, Engines, parts, components, equipment installed thereon or appurtenant thereto, free and clear of all liens, claims, charges and encumbrances other than Permitted Liens, including any mortgage, pledge, lien, charge, assignment, hypothecation, security interest, lease, title retention, preferential right or trust agreement, pooling, exchange, overhaul or repair agreement, or any other interest affecting title, and that Seller will warrant and defend such title against all claims and demands whatsoever.

EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES MADE BY SELLER SET FORTH IN THE AIRCRAFT SALE AND PURCHASE AGREEMENT, THE AIRCRAFT (AS DEFINED) IS BEING SOLD AND DELIVERED TO PURCHASER "AS IS" AND "WHERE IS", AND WITHOUT ANY REPRESENTATION, GUARANTEE OR WARRANTY OF SELLER EXPRESS OR IMPLIED, OF ANY KIND, ARISING BY LAW OR OTHERWISE.

WITHOUT LIMITING THE GENERALITY OF THE ABOVE, EACH OF PURCHASER AND JETRAN UNCONDITIONALLY AGREES THAT EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES MADE BY SELLER SET FORTH IN THE AIRCRAFT SALE AND PURCHASE AGREEMENT, THE AIRCRAFT (AS DEFINED) IS TO BE SOLD AND PURCHASED IN AN "AS IS", 'WHERE IS" CONDITION AS AT THE DELIVERY DATE, AND NO TERM, CONDITION, WARRANTY, REPRESENTATION OR

COVENANT OF ANY KIND HAS BEEN ACCEPTED, MADE OR IS GIVEN BY SELLER OR ITS SERVANTS OR AGENTS IN RESPECT OF THE AIRWORTHINESS, VALUE, QUALITY, DURABILITY, DATE PROCESSING, CONDITION, DESIGN, OPERATION, DESCRIPTION, MERCHANTABILITY OR FITNESS FOR USE OR PURPOSE OF THE AIRCRAFT OR ANY PART THEREOF, AS TO THE ABSENCE OF LATENT, INHERENT OR OTHER DEFECTS (WHETHER OR NOT DISCOVERABLE), AS TO THE COMPLETENESS OR CONDITION OF THE AIRCRAFT DOCUMENTS, AND/OR AS TO THE ABSENCE OF ANY INFRINGEMENT OF ANY PATENT, COPYRIGHT, DESIGN, OR OTHER PROPRIETARY RIGHTS. ALL TERMS, CONDITIONS, WARRANTIES AND REPRESENTATIONS (OR OBLIGATION OR LIABILITY, IN CONTRACT OR IN TORT) IN RELATION TO ANY ONE OR MORE

OF THOSE MATTERS, EXPRESSED OR IMPLIED, STATUTORY OR OTHERWISE1 ARE EXPRESSLY EXCLUDED.

This Aircraft Bill of Sale is made and delivered pursuant to the provisions of the Aircraft Sale and Purchase Agreement and is governed by the laws of the State of New York.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY L~FT BLANK]

	IN	WITNESS	WHEREOF	Seller	has	caused	this	instrument	to	be	executed	bу	its
duly	aut	chorized	represer	ntative	on _			2005.					

SELLER:

JETRAN, LLC

By:

Name:

Title:

SCHEDULE 7
FORM OF ACCEPTANCE CERTIFICATE

## One (1) B737-200 Aircraft

manufacturer's serial number

(as more particularly defined in the Aircraft Sale and Purchase Agreement described below, the "Aircraft")

(the "Purchaser") hereby certifies that pursuant to the Aircraft Sale & Purchase Agreement dated as of August .2005 between Jetran, LLC, as seller (the "Seller') and JetGlobal, LLC, as Purchaser (the "Aircraft Sale and Purchase Agreement'):

- (a) Purchaser has inspected the Aircraft, and the Aircraft conforms with the description and is in the condition and equipped as required by the Aircraft Sale and Purchase Agreement;
- b) Purchaser has accepted delivery of the Aircraft and is fully satisfied with the Aircraft and such acceptance;
- (c) Purchaser has received and inspected all of the Aircraft Documents (as defined in the Aircraft Sale and Purchase Agreement) and found them to be complete and satisfactory;

Purchaser acknowledges that it has no rights or claims whatsoever against Seller in respect of the condition of the Aircraft or the Aircraft Documents or any of the other matters referred to in Clause 8.4 of the Aircraft Sale and Purchase Agreement.

Date:

Duly executed for the Purchaser by;

JETGLOBAL, LLC

By:

Name:

Title:

SCHEDULE 8

[CONFIDENTIAL PORTION DELETED AND FILED SEPARATELY WITH THE SECURITIES AND EXCHANGE COMMISSION]

		SCH	HEDUI	LE 9	
FORM	ΟF	ASSIGNMENT	ΔND	MOTTAMIISS	ACREEMENT

ASSIGNMENT AND ASSUMPTION OF OWNERSHIP INTEREST  $[N\_\_\_]$ 

The form of the Assignment and Assumption Agreement between Seller and Purchaser shall be in the same form as set forth in Exhibit A of the Omnibus Agreement between

Seller and GECAS